# DISASTER INSURANCE OR A DISASTROUS INSURANCE – NATURAL DISASTER INSURANCE IN FRANCE

## MARIO JAMETTI THOMAS VON UNGERN-STERNBERG

## CESIFO WORKING PAPER NO. 1303 CATEGORY 2: PUBLIC CHOICE October 2004

An electronic version of the paper may be downloaded• from the SSRN website:www.SSRN.com• from the CESifo website:www.CESifo.de

# DISASTER INSURANCE OR A DISASTROUS INSURANCE – NATURAL DISASTER INSURANCE IN FRANCE

## Abstract

We model natural disaster insurance in France. We explicitly take into account the main institutional features of the system, such as the uniform premium rate in both high and low risk regions and the existence of a state reinsurance company. Our model indicates that the institutional set-up is fundamentally flawed. We find that the market is likely to lead to "specialist" equilibria, where insurers specialize in serving either high or low risk regions. As a result the reinsurance company, which offers cover to all insurers at the same price, is likely to suffer from a portfolio with mainly "bad" risks. We show that increasing the premium rate customers have to pay, a policy undertaken by the French authorities, will not necessarily solve these problems and comes at a high cost to the final consumer (and taxpayer).

JEL Code: D78, G22, L11.

Keywords: property insurance, reinsurance, risk selection.

Mario Jametti McMaster University Department of Economics KTH 439, 1280 Main Street West Hamilton, Ont. L8S 4M4 Canada jametti@mcmaster.ca Thomas von Ungern-Sternberg École des HEC University of Lausanne 1015 Lausanne Switzerland Thomas.vonUngern-Sternberg@hec.unil.ch

We wish to thank Marius Brülhart, François Marechal, Deszö Szalay and Ernst-Ludwig von Thadden for helpful comments. Remaining errors and omissions are ours.

## 1 Introduction

The recent evolution of climatic conditions in many parts of the world has brought insurance against natural disasters back on the policy agenda. Both the costs of floods and the violence of storms (e.g. the storm Lothar at the end of 1999 has caused damage of several billion Euro in France, Germany and Switzerland) seem to have increased in the last decade. Natural disasters have specific characteristics that lead to market failure in the absence of government intervention. In particular the "objective" distribution of damage is very hard to specify (or varies substantially across time) and the distribution of claims is very uneven across space. In Britain for example, damage due to subsidence (unheard of prior to the 1970s) has caused claims of the order of £3.5 billion since 1975. Only areas with clay soils are affected, since this type of soil strongly contracts when it dries out.

Natural disasters are often classified as "uninsurable". Jaffee and Russell (1996) indicate three characteristics that are often mentioned in this context: i) the market displays problems of adverse selection and moral hazard, ii) the potential loss is very large and iii) it is difficult (impossible) to obtain precise estimates of the probability of the event.

In many countries, such particularities have led to some form of government intervention in the market for natural disaster insurance. Von Ungern-Sternberg (2004) examines the property insurance market in five countries (Britain, Spain, France, Germany and Switzerland). He shows that there is a wide variety of regulatory systems in these countries, ranging form integrated state insurance in some parts of Switzerland to laissez-faire in Britain.

In the present paper, we study the French system of natural disaster insurance. Its main specificity is the fact that it is a mixed system in which both the state and the private insurance sector play an important role. Let us start off by briefly outlining its basic structure<sup>1</sup>. Natural disaster insurance was introduced in France in 1982 as a reaction to a severe flood a few months earlier. The term "natural disaster" ("catastrophe naturelle") is not defined in the law creating the system. A commission, formed of representatives of the Ministries of Interior, Finance and Environment, has to decide whether a given occurrence is deemed a natural disaster and hence makes claimants eligible for reimbursement. The conditions of the insurance (e.g. premium rates and excesses) are fixed by decree and uniform across the country. Insurance is compulsory, presumably to reduce problems of adverse selection among property owners. Similarly, all insurance companies offering (other types of) property insurance in a specific area are obliged to include protection against natural disasters. Premium rates are defined as a percentage of other property insurance premiums (in particular fire), while excesses (for non-commercial buildings) are fixed amounts per contract and event.

An important institutional feature for our analysis is the existence of a publicly owned reinsurance company, the Caisse Centrale de Réassurance (CCR). Reinsurance is not compulsory, and insurers are free to contract with other, private, reinsurance companies. Reinsurance with the state reinsurance company is, however, particularly attractive, both because of the reinsurance premiums it charges and because it can offer unlimited cover (it is covered by a state guarantee). Insurance companies that decide to reinsure with the public reinsurer are offered two types of contracts; proportional contracts (for a given percentage of premium income the reinsurance company covers a given percentage of claims) and stop-loss contracts (the reinsurance company covers all claims that exceed a given multiple of annual premium income). The combination of these two types of reinsurance necessarily implies that the CCR (and ultimately the tax-payer) will bear most of the cost when a large-scale disaster occurs.

In the first 20 years of its existence, the CCR never managed to accumulate any substantial level of reserves. In 1999, it was on the verge of bankruptcy, and had to be recapitalized with FF 1 billion. All this occurred in spite of the fact that the claims/premium ratio for the entire period was only of the order of 55% (c.f. von Ungern (2004), p. 91). Parallel

 $<sup>^{1}</sup>$ The description below is drawn from von Ungern (2004). The author also provides a more in depth description of the natural disaster insurance in France.

to the recapitalization, premiums and excesses for the property owners were raised by 30%, and the amount of coverage the CCR was willing to supply on a proportional basis was limited to  $50\%^2$ .

The purpose of this paper is to build a simple model which allows us to understand why the reinsurance company was unable to participate in the profits of the system to any significant extent. In our model, two forces drive this result. First, there is a problem of regulatory capture. The CCR paid the French insurance companies a large proportion of premium to cover, mainly fictitious, "administrative costs".<sup>3</sup> Second, the CCR is subject to important problems of risk selection. Given the structure of the system, there is a natural tendency for certain insurers to concentrate only on the "good" regions and buy little reinsurance, while others insure the "bad" regions and then reinsure with the CCR.

Our model shows that increasing the premium rate might not be an adequate solution to this problem. Indeed, the higher the premium rate, the greater the incentives to "separate" the good from the bad risks. If one takes into account this adjustment effect, it is not clear that an increase in the customer premium rate will in fact solve the CCR's problems.

In a more general discussion we address the question of the full "social" cost of the system. This cost is defined as the sum of final costumer premium payments and expected losses of the reinsurance company. The latter presumably covered by the state. The model illustrates that increasing the premium rate to improve the financial situation of the reinsurance comes at a high cost to the taxpayer.

Within the simple model, we work with risk neutral insurers. This necessarily implies that the choice to reinsure is discrete (either full or no reinsurance). If expected damage is lower than the reinsurance premium the insurer will not buy reinsurance cover. If the inverse holds, it will buy complete cover, instead. This in turn implies that the reinsurance company cannot be financially viable (reinsurance premium income will be lower than expected cost). This property does not necessarily hold if one were to work with risk averse insurers. For this reason we introduce a more general (and more complex) variant of the model, where insurers are risk averse. This change results in "interior" solutions for reinsurance and potentially positive (expected) profits for the reinsurer. This more general model is no longer analytically tractable, but we conjecture that the outcome would be very similar to our basic model, especially the fact that an increase in the premium rate is likely to lead to a higher degree of risk selection.

As mentioned above, von Ungern (2004) provides a critical appraisal of the French natural disaster insurance. He describes empirically most of the problems studied in this paper. However, he does not proceed to a formal analysis. A more favourable assessment of the system is given in Michel-Kerjan (2001), concluding that "the partnership between public and private sectors has passed the test of twenty years of operation". This author does not address the problem of risk selection, and he assigns the depletion of reserves of the public reinsurance company (which lead to the collapse of the system in 1999) to an exceptional succession of high damage years in the 1990s. This assessment seems unconvincing, since, as shown in von Ungern (2004) using data from the CCR, the accumulated excess premium income (premium income minus damage payments) over the period 1990-1998 was over 10 billion French Francs. This amount alone would have been enough to pay the entire damage payments for the year 1999 of 6.6 billion French Francs.

The paper is organised as follows: Section 2 presents the model and characterizes potential equilibrium outcomes. Section 3 discusses which of the potential equilibria are Nash equilibria. Section 4 presents the effect of an increase in the premium rate. Section 5 addresses the situation of the reinsurance company, demonstrating that the effect from an increase in the premium on its expected profits is dampened with a higher degree of risk selection. Section 6 presents a discussion of our results, while Section 7 concludes.

<sup>&</sup>lt;sup>2</sup>Stop loss cover continues to be available.

 $<sup>^{3}</sup>$ In Spain, working with an identical system of premium collection, commission for administrative costs is 5%, while it was, until recently, 24% in France.

## 2 The model

#### 2.1 Setup

Consider a country with two regions  $i \in [H; L]$ . A region has a stock of  $\mathcal{H}$  houses, each of (normalized) value 1. We assume that a house can be affected by damage, which destroys the house completely (damage value = 1). Damage occurrence for each house in region i is assumed to follow a binomial distribution with the probability of damage equal to  $p_i$ . We assume that damage occurrence is independent both among houses and among regions. Hence, for each region the expected damage is  $\mathcal{H}p_i$ . Regions differ only in the respective probability of an event occurring, with the H-region having a higher probability of damage than the L-region  $(p_H > p_L)$ .

Housing insurance is compulsory, and its price is fixed at rate  $\theta$  independent of  $p_i$ . The service is provided by identical insurers in a competitive market. Therefore, the market outcome in the model will be determined by a symmetric, free-entry equilibrium. Whenever an insurer j decides to offer service in a particular region i, its market share will be given by  $1/N_i^*$ , where  $N_i^*$  is the number of insurers active in the region.<sup>4</sup>

There exists a single reinsurance company which offers proportional contracts to the market in all regions.<sup>5</sup> Reinsurance is voluntary, and each insurer can decide on the fraction of the portfolio that it wants to reinsure. We denote by  $r_j \in [0, 1]$  the rate of retention for insurer j.<sup>6</sup> Note that an insurer can reinsure only its entire portfolio and therefore can only choose one level of  $r_j$ , even when active in both regions.

The premium income from the fraction of the portfolio that is reinsured is divided as follows. The insurer keeps a percentage  $\gamma$  to cover administrative costs and passes  $(1 - \gamma)$  to the reinsurance company. For the remaining portfolio, the insurer keeps the entire premium income. The reinsurer covers the fraction  $(1 - r_j)$  of claims payments. In terms of a particular house in region *i*, revenue for insurer *j* from signing a natural disaster insurance contract can be expressed by the following random variable:

$$Rev_{j}(i) = \begin{cases} \theta\gamma + r_{j}\theta(1-\gamma) \\ \theta\gamma + r_{j}\left[\theta(1-\gamma) - 1\right] & with \, prob. \quad \begin{pmatrix} 1-p_{i} \end{pmatrix} \\ p_{i} \end{cases}$$

The insurer is certain to obtain  $\theta\gamma$  for each house it has under contract. Further, with probability  $(1 - p_i)$  the house does not suffer a damage. The insurer then receives the remaining premium income for the fraction of retained risks  $(r_j)$ . With probability  $p_i$  the house is destroyed (damage value equals 1) and "net revenue" to the insurer is the fraction of remaining premium income minus the fraction of damage that the insurance company bears.

All costs other than claims payments are assumed to be fixed. However, we consider two types of fixed costs. A region specific fixed cost (f) and a country specific fixed cost (F). F represents the fact that some costs, such as setting up a countrywide representation, must be incurred independently of the number of regions covered. Other fixed costs, such as local branch activities, are specific to a particular region. For simplicity we assume that regional fixed costs (f) are the same in both regions.<sup>7</sup> The country specific fixed cost (F) plays a crucial role in this model. Without it, there is no incentive for an insurer to serve both regions. In fact, in the absence of F and under a competitive insurance market characterized by free entry, each individual insurer will seek a homogeneous risk portfolio in order to choose an optimal rate of retention. Only through the introduction of country fixed costs might an insurer actually want to provide service in both regions. We will show below

 $<sup>^{4}</sup>$ We work with free-entry equilibria to make the model simple. A high N in the model represents the substantial profits earned by the French insurers.

<sup>&</sup>lt;sup>5</sup>This is an exact model of the proportional reinsurance cover the CCR offered prior to 2001.

<sup>&</sup>lt;sup>6</sup>Hence  $(1 - r_j)$  is the fraction of the portfolio that is reinsured.

<sup>&</sup>lt;sup>7</sup>One could argue that administrative costs in the H-regions are higher since there will be, on average, more cases to evaluate and forms to fill out. We abstract from this.

that the French institutional setup leads to a tendency for risk selection despite country fixed costs.

There can be two types of insurers in our model. A specialist (insurer) serves only one specific region. Specialist j's expected profit if serving region i is

$$E[\pi_j] = \frac{\mathcal{H}}{N_i^*} \left[\theta\gamma + r_j \left(\theta \left(1 - \gamma\right) - p_i\right)\right] - f - F,\tag{1}$$

where  $N_i^*$  is the total number of insurers (of both types) operating in region *i*. Similarly, a *generalist (insurer)* serves both regions, with expected profits given by:

$$E\left[\pi_{j}\right] = \frac{\mathcal{H}}{N_{L}^{*}}\left[\theta\gamma + r_{j}\left(\theta\left(1-\gamma\right)-p_{L}\right)\right] + \frac{\mathcal{H}}{N_{H}^{*}}\left[\theta\gamma + r_{j}\left(\theta\left(1-\gamma\right)-p_{H}\right)\right] - 2f - F.$$
 (2)

Each insurer is assumed to be risk neutral, and hence will choose  $r_j$  to maximize expected profits and decide to enter as long as expected profits are positive. This implies that there can be three candidates of market equilibria.

**Definition 1** i) in a specialist equilibrium (SE), all insurers serve only one region. Such an equilibrium is characterized by  $SE(r_L; r_H)$ , where  $r_i$  is the rate of retention for specialists operating in region i.

ii) in a generalist equilibrium (GE), all insurers serve both regions. Such an equilibrium is characterized by  $GE(r_G)$ .

iii) in a hybrid equilibrium (HE), specialist and generalist insurers coexist. Such an equilibrium is characterized by  $HEL(r_S; r_G)$  if the specialists operate in the L-region and by  $HEH(r_S; r_G)$  if the specialists operate in the H-region.  $r_S$  is the rate of retention of specialists and  $r_G$  the one for generalists.

Let us start off by looking at the optimal choice for  $r_j$  for the insurance companies. The expression for expected profits is linear in  $r_j$ , for both the specialist (1) and the generalist (2) characterization. Hence, the optimal choice of retention will be either zero or one. Specifically, from (1) we have for a specialist insurer:

$$\begin{aligned} r_j &= 1 & if \ \theta \left( 1 - \gamma \right) - p_i \geq 0 \\ r_i &= 0 & otherwise \end{aligned}$$
 (3)

Note that the choice of retention is independent of the number of firms in the region. Similarly, using (2) we see that for a generalist insurer we have:

$$r_{j} = 1 \quad if \quad \frac{\mathcal{H}}{N_{L}^{*}} \left( \theta \left( 1 - \gamma \right) - p_{L} \right) + \frac{\mathcal{H}}{N_{H}^{*}} \left( \theta \left( 1 - \gamma \right) - p_{H} \right) \ge 0$$
  

$$r_{j} = 0 \quad otherwise \qquad (4)$$

In this case the optimal rate of retention depends on the number of insurers, since it is potentially different across regions. We next turn to each of the equilibrium candidates.

#### 2.2 Equilibrium candidates

Equilibrium candidates are characterized by two factors: the optimal choice of the rate of retention for each insurer and the number of each type of insurers, which is obtained by setting the corresponding profit, (1) or (2), equal to zero. Table 1 presents the eight equilibrium candidates and Appendix A provides more detail on how these candidates are obtained.

We observe that there exist three candidates for specialist equilibria (SE), one where specialists in both regions fully reinsure, one where only the insurers in the *H*-region reinsure and one with no reinsurance in both regions. The number of insurers in each market depends on the equilibrium rate of retention. Similarly, there are two equilibrium candidates for generalist equilibria, no reinsurance or full reinsurance. Again the number of insurer varies according to the rate of retention. Finally, there are three candidates of hybrid equilibria:<sup>8</sup> one where generalists fully reinsure and specialists, without reinsuring, operate in the Lregion; one with specialists in the *L*-region but no reinsurance for either type of firms; and one with specialists, fully reinsuring, in the *H*-region and generalists not reinsuring. In addition to the number of generalists, Table 1 displays the total number of insurers in each of the hybrid equilibrium candidates. The number of specialists can be obtained by difference.

Table 1		
Equilibrium	candidates	

Familibuium	Number of ins.	Number of ins.	Number of
Equinorium	in $L$ -region	in $H$ -region	generalists
$SE(r_L = 0; r_H = 0)$	$rac{\mathcal{H} heta\gamma}{(f\!+\!F)}$	$rac{\mathcal{H} heta\gamma}{(f+F)}$	
$SE(r_L = 1; r_H = 0)$	$rac{\mathcal{H}( heta - p_L)}{(f + F)}$	$rac{\mathcal{H} heta\gamma}{(f\!+\!F)}$	
$SE(r_L = 1; r_H = 1)$	$rac{\mathcal{H}(\theta - p_L)}{(f + F)}$	$rac{\mathcal{H}( heta - p_H)}{(f + F)}$	
$GE\left(r_G=0\right)$			$rac{2\mathcal{H} heta\gamma}{(2f+F)}$
$GE\left(r_G=1\right)$			$\frac{\mathcal{H}(\theta - p_L + \theta - p_H)}{(2f + F)}$
$HEL\left(r_S=1;r_G=0\right)$	$\frac{\mathcal{H}(\theta - p_L)}{(f + F)}$		$\frac{\mathcal{H}\theta\gamma}{f\left(2-\frac{\theta\gamma}{\theta-p_L}\right)+F\left(1-\frac{\theta\gamma}{\theta-p_L}\right)}$
$HEL(r_S=1;r_G=1)$	$\frac{\mathcal{H}(\theta - p_L)}{(f + F)}$		$\frac{\mathcal{H}(\theta - p_H)}{f}$
$HEH\left(r_{S}=0;r_{G}=1\right)$		$rac{\mathcal{H} heta\gamma}{(f+F)}$	$\frac{\mathcal{H}(\theta - p_L)}{f\left(2 - \frac{\theta - p_H}{\theta\gamma}\right) + F\left(1 - \frac{\theta - p_H}{\theta\gamma}\right)}$

Appendix A presents comparative statics results of the number of insurance companies in each equilibrium candidate with respect to the parameters of the model. Generally, the number of houses  $(\mathcal{H})$ , the premium rate  $(\theta)$  and the commission rate  $(\gamma)$  increase the number of insurers, while the damage probabilities  $(p_i)$  and the fixed costs (f, F) decrease the number of them. There are some exceptions to this pattern (as shown in Appendix A), mainly for the hybrid equilibrium candidates, where a change in a parameter also implies a redistribution between the types of insurers, and thus the comparative static effects can be ambiguous.

#### 3 Nash equilibria

#### Equilibrium conditions 3.1

Having described equilibrium candidates we now determine under which conditions these are Nash equilibria.<sup>9</sup>

For this we explore whether, and under which conditions, a single insurer has an incentive to deviate from its candidate equilibrium strategy, under the assumption that all the other insurers play the equilibrium strategies. Note that the set of possible deviations is not very large. Retiring from the market is never a profitable deviation, since equilibria

<sup>&</sup>lt;sup>8</sup>Appendix A explains why other potential characterizations such as  $HEL(r_S = 0; r_G = 0)$  are in fact not equilibrium candidates. <sup>9</sup>Any equilibria we describe below are Nash.

are characterized by zero expected profits. Deviation strategies are defined by entering or retiring from a particular region choosing the optimal rate of retention.<sup>10</sup> In particular, we have the following deviation strategies: i) in a specialist equilibrium the deviating insurer will want to serve both regions, ii) in a generalist equilibrium the deviating insurer will specialize in the *L*-region and iii) in a *HEL* a generalist will deviate by specializing into the *H*-region, while in a *HEH* a generalist will deviate by specializing into the *L*-region. Appendix B provides a detailed description of these equilibrium conditions. The results are displayed in Table 2.

The second column in Table 2 indicates the rate of retention  $(r_D)$  chosen by the deviating insurer. Note that in most equilibrium candidates this choice is uniquely determined by the conditions which must hold for the equilibrium candidate. Only in the  $SE(r_L = 1; r_H = 0)$  and  $GE(r_G = 0)$  is it possible that the rate of retention  $r_D$  can be both zero or one, depending on parameter values. The last two columns indicate the conditions under which deviation is not profitable for each of the regions. Thus, for the specialist equilibrium candidates there is a condition that must hold in each region. In the generalist equilibrium, since deviation is into the *L*-region, the condition must only hold in the *L*-region. Similarly, for the *HEL* candidates deviation is to specialize into the *H*-region, thus the condition must hold in this region and inversely for the *HEH* candidate.

For the sake of completeness, Table 2 also includes conditions that must hold such that at least one insurer is active in the market.

To sum up, three conditions must hold for a candidate to be an equilibrium. First, the choice of the rate of retention of the candidate must indeed be optimal according to (3) and (4). Second, the corresponding equilibrium condition displayed in Table 2 must hold. Third, the condition that at least one insurer is active in the market must also hold.

#### 3.2 Areas of equilibrium

We can describe these equilibrium conditions graphically. This tool will also help us to illustrate the analysis of an increase in the premium rate. For this we introduce the concepts of "switch" and "jump" lines.

**Definition 2** Switch lines are defined by situations where either (3) or (4) hold with equality. They are represented by ri in the graphs, where i corresponds to the type of firm.

Jump-lines are defined by a situation such that a particular equilibrium condition holds with equality. These lines are represented by  $J(r_H; r_L; r_D)$  for specialist equilibria,  $J(r_G; r_D)$ for generalist equilibria and  $J(r_S; r_G; r_D)$  for hybrid equilibria.

Intuitively, switch lines indicate where an individual insurer will "switch" regime from zero to full retention. Similarly, jump lines indicate when a single insurer will "jump" from the actual equilibrium configuration. In a broader sense, by crossing these lines we change from one equilibrium characterization to another.

These graphical devices allow us to illustrate the area of equilibria for particular parameter values, defined by the area where a single insurer does neither want to switch retention regime nor wants to jump to another equilibrium configuration. For the graphical analysis, and without loss of generality, we define  $p_H$  and  $p_L$  in terms of a mean preserving spread such that:

$$\overline{p} = \frac{p_H + p_L}{2}$$

 $<sup>^{10}</sup>$ Note that the set of relevant deviation strategies is finite due to risk neutrality, even though the strategy space is not.

Equilibrium Candidates	$r_D$	Nash-equilibrium condition in <i>L</i> -region or cond. on minimum number of insurers	Nash-equilibrium condition in $H$ -region
$SE(r_L = 0; r_H = 0)$	0	$\frac{\mathcal{H}\theta\gamma}{\mathcal{H}\theta\gamma+f+F}\left(f+F\right)-f\leq0$	$\frac{\mathcal{H}\theta\gamma}{\mathcal{H}\theta\gamma+f+F}\left(f+F\right)-f\leq0$
$SE(r_L = 1; r_H = 0)$ Case 1	0	$\begin{aligned} f\left[\frac{\theta\gamma}{(\theta-p_L)} + \frac{\mathcal{H}\theta\gamma}{\mathcal{H}\theta\gamma+f+F} - 2\right] \\ + F\left[\frac{\theta\gamma}{(\theta-p_L)} + \frac{\mathcal{H}\theta\gamma}{\mathcal{H}\theta\gamma+f+F} - 1\right] \leq 0 \end{aligned}$	$\frac{\mathcal{H}\theta\gamma}{\mathcal{H}(\theta-p_L)+f+F}\left(f+F\right)-f\leq 0$
$SE(r_L = 1; r_H = 0)$ Case 2	1	$\frac{\mathcal{H}(\theta - p_H)}{\mathcal{H}\theta\gamma + f + F} \left( f + F \right) - f \le 0$	$f\left[\frac{\mathcal{H}(\theta-p_L)}{\mathcal{H}(\theta-p_L)+f+F} + \frac{(\theta-p_H)}{\theta\gamma} - 2\right] + F\left[\frac{\mathcal{H}(\theta-p_L)}{\mathcal{H}(\theta-p_L)+f+F} + \frac{(\theta-p_H)}{\theta\gamma} - 1\right] \le 0$
$SE(r_L = 1; r_H = 1)$	1	$\frac{\mathcal{H}(\theta - p_H)}{\mathcal{H}(\theta - p_H) + f + F} \left( f + F \right) - f \le 0$	$\frac{\mathcal{H}(\theta - p_L)}{\mathcal{H}(\theta - p_L) + f + F} \left( f + F \right) - f \le 0$
$GE (r_G = 0)$ Case 1	0	always Nash	
$\begin{array}{l} GE\left(r_{G}=0\right)\\ \text{Case }2 \end{array}$	1	$f\left[\frac{(\theta-p_L)}{\theta\gamma}-1\right]+F\left[\frac{(\theta-p_L)}{2\theta\gamma}-1\right] \le 0$	
$GE\left(r_G=1\right)$	1	$f\left[\frac{2(\theta-p_L)}{(\theta-p_L+\theta-p_H)}-1\right] + F\left[\frac{(\theta-p_L)}{(\theta-p_L+\theta-p_H)}-1\right] \le 0$	
$HEL\left(r_S=1;r_G=0\right)$	0		$f - \frac{\theta \gamma}{(\theta - p_L)} \left( f + F \right) \le 0$
$HEL\left(r_S=1;r_G=1\right)$	1		always Nash
$HEH\left(r_{S}=0;r_{G}=1\right)$	1	$f - \frac{(\theta - p_H)}{\theta \gamma} \left( f + F \right) \le 0$	
Min. number in SE		$\mathcal{H} heta\gamma \ge (f+F)$	
$\begin{array}{c} \text{Min. number} \\ \text{in } GE \end{array}$		$\mathcal{H} heta\gamma \geq rac{(2f+F)}{2}$	
Min. number in <i>HE</i>		$\mathcal{H}\theta\gamma \ge 2(f+F)$	

 Table 2

 Summary of Nash equilibrium and minimum number of insurer conditions

Figure 1 illustrates the area of different types of equilibria under a set of parameter values in the  $(\theta, p_H)$ -space.<sup>11</sup> Each graph includes the relevant jump- and switch-lines for a given equilibrium candidate. These lines, together with the minimum-number-of-firms condition determine the (shaded) area of equilibrium. A more detailed description of the graphical analysis can be found in Appendix B.

We observe that a  $GE(r_G = 0)$  equilibrium can only exist for low premium rates and rather small differences in the damage probability.<sup>12</sup> Similarly, a  $GE(r_G = 1)$  is sustained with high premium rates and small differences in the damage probabilities. Thus, as mentioned above, the introduction of country specific fixed costs is not sufficient to generate an outcome of insurers offering in both regions. In fact, the cost of not being able to adjust the rate of retention optimally when offering in both regions outweighs the benefit of "diluting" the country fixed cost, at least for intermediate and high differences in damage probabilities across the regions.

Quite intuitively, the opposite occurs for the  $SE(r_L = 1; r_H = 0)$ . This equilibrium will be sustained in cases where there are important differences in the damage probabilities and for intermediate values of the premium rate. In such cases, insurers prefer to have a homogeneous risk portfolio, even though they have to support the full country fixed cost.

The  $HEL(r_S = 1; r_G = 0)$  is an equilibrium for intermediate values of both the premium rate and the difference in damage probabilities, while the  $HEL(r_S = 1; r_G = 1)$  can be sustained both with low and high differences in the damage probabilities for sufficiently high premium rates. Finally, a  $HEH(r_S = 0; r_G = 1)$  requires intermediate premium rates and rather important differences in the damage probabilities.

Appendix B also presents a sensitivity analysis of the equilibrium areas with respect to the relative weight of the two fixed costs. Quite intuitively, we observe that relatively high country fixed costs expand the equilibrium areas of the generalist and hybrid equilibria, while relatively high regional fixed costs expand the area of the  $SE(r_L = 1; r_H = 0)$  and (for the parameter values chosen) make a  $SE(r_L = 0; r_H = 0)$  possible.

Figure 2 presents the different equilibrium areas jointly. To reduce the number of areas, the area for a  $GE(r_G = 1)$  has been omitted, as well as the condition on the minimum number of firms.

<sup>&</sup>lt;sup>11</sup>The types of equilibria not displayed in Figure 1 do not exist under the chosen set of parameter values. <sup>12</sup>Note that, due to the specification of a mean preserving spread of  $p_L$ ,  $p_H$  around  $\overline{p}$ , a high  $p_H$  in the graph implies a low  $p_L$ .

Figure 1 Areas of equilibrium



Parameter values:  $H = 100,000; \delta = 0.2; f = 200; F = 200; \overline{p} = 0.05.$ 

Figure 2 Joint areas of equilibrium



Parameter values:  $H = 100,000; \delta = 0.2; f = 200; F = 200; \overline{p} = 0.05.$ 

We see that for each point in the graph there exists at least one equilibrium. In fact, our model satisfies the conditions of Proposition 8.D.3 of Mas-Colell, Whinston and Green (1995) and hence we have existence of a pure strategy equilibrium. Further we observe that some equilibrium areas overlap. Particularly the  $HEL(r_S = 1; r_G = 0)$  is embedded in the  $SE(r_L = 1; r_H = 0)$  area and the  $GE(r_G = 0)$  area. The intuition behind this overlap is that the number of insurers in the market can be such that one or the other equilibrium occurs. For example, in the area of overlap of the  $HEL(r_S = 1; r_G = 0)$  and the  $GE(r_G = 0)$  we have that the number of insurers is higher in the generalist equilibrium. Loosely speaking, we will end up with a generalist equilibrium if "many" insurers decide to enter the market.

#### 3.3 Situation under redistributive premium rates

We next briefly explore the outcome of our model if the premium rate is chosen with a goal of redistribution between high and low probability regions. The feature of a unique premium rate within the country, without distinguishing different, regional risk exposition is a clear indication that the French legislator willingly imposed a certain redistribution among high and low risk areas. It is therefore informative to analyze what the outcomes are in our model when the low risk areas subsidize damage payments in more exposed areas. Formally, we assume  $p_L < \theta < p_H$  to represent a redistributive premium rate. Further, we assume that the premium rate is chosen such that

$$\theta \left(1-\gamma\right) - p_L + \theta \left(1-\gamma\right) - p_H \ge 0.$$

This additional assumption ensures that if all houses where reinsured the reinsurance company would at least break even.

In such a situation only few of the candidates are actually potential equilibria. First of all, using (3) and the assumptions on the redistributive premium rate, we can observe that for specialist insurers:  $r_L = 1$  and  $r_H = 0$ . This directly eliminates the  $SE(r_L = 0; r_H = 0)$  and  $SE(r_L = 1; r_H = 1)$  as equilibrium candidates. Further, due to  $\theta < p_H$  we have that the equilibrium conditions for  $SE(r_L = 1; r_H = 0)$  Case 2 are always satisfied. Finally, conditions in  $SE(r_L = 1; r_H = 0)$  Case 1 remain unchanged.

Regarding the generalist equilibrium candidates we observe that our assumptions on the redistributive premium rate implies that only  $GE(r_G = 1)$  is a valid candidate. However, in that case the equilibrium condition is never satisfied, such that no generalist equilibrium exists.

Lastly, the only hybrid equilibrium characterization that is a potential equilibrium is  $HEL(r_S = 1; r_G = 0)$ . The other characterizations are ruled out by redistributive premium rates.

**Remark 3** Under the realistic assumption of redistributive premium rates across the country, we observe that only a few of our equilibrium candidates are actually potential equilibria. Note especially that no generalist equilibrium is possible under redistributive premium rates. Thus, the model clearly illustrates the tendency towards risk selection of the institutional setup.

### 4 Effect of an increase in the premium rate

The effect of an increase in the premium rate should not be summarized in a simple comparative statics exercise. In general, (marginally) increasing the premium rate would not alter the situation in the market. As shown in Appendix A, increasing the premium rate would in our model lead to an increased number of firms under each specific type of equilibrium. Much more interestingly, discrete increases in  $\theta$  might provoke dramatic changes in the market structure in the sense that the type of insurers serving each region might change. From the graphical analysis above we have seen that changes in the premium rate can easily lead to a situation where the situation after the change is a different type of equilibrium. Analyzing discrete changes are in line with the decision of the French authorities when increasing the premium rate by 30%.

Our graphical tools are useful for this analysis. We describe sequences of the equilibrium outcome for different levels of the premium rate. Naturally, one can think of several different initial situations. We chose to examine sequences starting from a generalist equilibrium with full reinsurance and varying degrees of differences in damage probabilities across regions. In such a situation, three potential sequences can be found.<sup>13</sup> These sequences are illustrated in Figure 3.<sup>14</sup>

**Proposition 4** Starting from a generalist equilibrium with full reinsurance, the effect of an increase in the premium rate leads to the following sequences:

i) small difference in damage probabilities  $\implies GE(r_G = 0) \rightarrow GE(r_G = 1)$ .

ii) intermediate difference in damage probabilities  $\implies GE(r_G = 0) \rightarrow$ 

 $HEL(r_S = 1; r_G = 0) \rightarrow SE(r_L = 1; r_H = 0) \rightarrow HEL(r_S = 1; r_G = 1).$ 

*iii)* large difference in damage probabilities  $\implies GE(r_G = 0) \rightarrow SE(r_L = 1; r_H = 0) \rightarrow HEL(r_S = 1; r_G = 1).$ 

<sup>&</sup>lt;sup>13</sup>Theoretically there could be a sequence involving a jump into a  $HEH(r_S = 0; r_G = 1)$ . However, as shown above, deviation from a GE implies specialization into the *L*-region, hence such a sequence could not be triggered by the deviation of a *single* insurer.

 $<sup>^{14}</sup>$  The areas of equilibrium indicated in the graph do not necessarily correspond to the full area of each type of equilibrium.

Figure 3 Equilibrium Sequences with increase in premium rate



Parameter values:  $H = 100,000; \delta = 0.2; f = 200; F = 200; \overline{p} = 0.05.$ 

Starting from a GE with small differences in the damage probabilities across the high and low regions, an increase in the premium rate corresponds to moving up along the vertical line starting from point A. We observe that a sufficient increase in the premium rate implies a switch in the retention regime (point B).

In the case of intermediate differences in damage probabilities, the initial starting equilibrium can be illustrated by point C. A sufficient increase implies that we jump to a  $HEL(r_S = 1; r_G = 0)$ , illustrated by point D. Choosing an even higher premium rate implies a specialist equilibrium of the  $SE(r_L = 1; r_H = 0)$  type (point E). Finally, with even further increases we end up in a  $HEL(r_S = 1; r_G = 1)$  (point F).

In the case of large differences in damage probabilities, the sequence is similar to that described above. The difference is that the passage via the  $HEL(r_S = 1; r_G = 0)$  area is bypassed. In this case the market configuration changes directly from a generalist to a specialist equilibrium (passage from point G to H). Further increases of the premium rate imply again a  $HEL(r_S = 1; r_G = 1)$  (point I).

**Remark 5** Starting from a generalist equilibrium configuration with a low premium rate, we observe that only one of the three sequences following increases in the premium rate involves no risk selection. In fact, only with small differences of damage probabilities can generalist equilibria be sustained for all choices of the premium rate. As soon as the differences in probabilities exceed a certain threshold, sufficient increases in the premium rate imply partial or full risk selection. Furthermore, we observe that neither sequence ii) nor iii) ever return to a generalist equilibrium.

Obviously, this result has important consequences for the financial situation of the reinsurance company. An increase in the premium rate does not necessarily leave the market structure unchanged, and thus improve the situation of the reinsurer. The implied changes in market structure might actually make the situation worse.

## 5 Profits of the reinsurance company

In our model, the position of the reinsurer is obviously very weak. It is a direct result of risk neutral insurers that expected profits of the reinsurance company are negative in equilibrium. This (unrealistic) property would obviously change if one were to work with risk averse insurers.<sup>15</sup> Bearing this in mind, one can nevertheless gain useful insights from our model regarding the relative change in expected profits with changes in the premium rate.

Table 3 indicates the expressions for expected profits in decreasing order of expected profits for the reinsurance company in the different potential equilibria.

Equilibria	Expected profit $(\pi_R)$
$SE(r_L = 1; r_H = 1), GE(r_G = 1), HEL(r_S = 1; r_G = 1)$	0
$SE(r_L = 0; r_H = 0), GE(r_G = 0)$	$\mathcal{H}\left[\left(\theta\left(1-\gamma\right)-p_{L}\right)+\left(\theta\left(1-\gamma\right)-p_{H}\right)\right]<0$
$HEL\left(r_S=1;r_G=0\right)$	$\frac{\frac{N_G}{N_S + N_G} \mathcal{H} \left( \theta \left( 1 - \gamma \right) - p_L \right)}{+ \mathcal{H} \left( \theta \left( 1 - \gamma \right) - p_H \right) < 0}$
$HEH\left(r_S=0;r_G=1\right)$	$\frac{N_S}{N_S + N_G} \mathcal{H} \left( \theta \left( 1 - \gamma \right) - p_H \right) < 0$
$SE(r_L = 1; r_H = 0)$	$\mathcal{H}\left(\theta\left(1-\gamma\right)-p_{H}\right)<0$

 Table 3

 Expected profits of the reinsurance company

 $<sup>^{15}\</sup>mathrm{See}$  the next section for a discussion.

Three equilibria  $(SE(r_L = 1; r_H = 1), GE(r_G = 1), HEL(r_S = 1; r_G = 1))$  have no role for the reinsurer. Next, under the  $GE(r_G = 0)$  and  $SE(r_L = 0; r_H = 0)$  we have full reinsurance, and hence the reinsurer obtains both the "good" and the "bad" risks. Under a  $HEL(r_S = 1; r_G = 0)$  the reinsurer also gets both types of risks. However, in this case only a fraction of the houses in the low probability region are reinsured. Finally, in the  $HEH(r_S = 0; r_G = 1)$  and  $SE(r_L = 1; r_H = 0)$  the reinsurer gets only "bad" risks. While it is only a fraction of bad risks in the  $HEH(r_S = 0; r_G = 1)$  it is the full portfolio of high damage houses in the  $SE(r_L = 1; r_H = 0)$ .

Table 4 presents the comparative statics with respect to the premium rate.

Equilibria	$\frac{\partial \pi_R}{\partial \theta}$
$ SE(r_L = 1; r_H = 1), GE(r_G = 1), HEL(r_S = 1; r_G = 1) $	0
$SE(r_L = 0; r_H = 0), GE(r_G = 0)$	$2\mathcal{H}\left(1-\gamma\right) > 0$
$HEL\left(r_S=1;r_G=0\right)$	$\frac{\frac{\partial \left(\frac{N_G}{N_S+N_G}\right)}{\partial \theta}}{\mathcal{H}} \mathcal{H}\left(\theta\left(1-\gamma\right)-p_L\right) + \frac{N_G}{N_S+N_G} \mathcal{H}\left(1-\gamma\right) + \mathcal{H}\left(1-\gamma\right) > 0$
$HEH\left(r_{S}=0;r_{G}=1\right)$	$\frac{\partial \left(\frac{N_S}{N_S + N_G}\right)}{\partial \theta} \mathcal{H} \left(\theta \left(1 - \gamma\right) - p_H\right) \\ + \frac{N_S}{N_S + N_G} \mathcal{H} \left(1 - \gamma\right) > 0$
$SE(r_L = 1; r_H = 0)$	$\mathcal{H}\left(1-\gamma\right) > 0$

 Table 4

 Comparative statics with respect to the premium rate

We note that, not considering eventual changes in the equilibrium characterization, a (marginal) increase in the premium rate decreases expected losses for the reinsurance company.<sup>16</sup> There is in fact an ordering in the magnitude of the derivative. We can show that

$$\frac{\partial \pi_R}{\partial \theta} \bigg|_{GE(r_G=0)} > \frac{\partial \pi_R}{\partial \theta} \bigg|_{HEL(r_S=1;r_G=0)} > \frac{\partial \pi_R}{\partial \theta} \bigg|_{SE(r_L=1;r_H=0)} .^{17}$$

This ordering is intuitive. When the reinsurer has the full portfolio, the effect of an increase in the premium rate is strongest. For equilibrium outcomes with more and more risk selection, the reinsurer "loses" part or all of the good risks, and thus the effect of an increase in the premium rate becomes less effective. In other words, *if increasing the premium rate aims at improving the situation of the reinsurer, then this policy is less effective in situations with a high degree of risk selection.* 

As mentioned above, a comparative statics exercise does not give us the full picture of the effect of an increase in the premium rate. One must also include the effect of the changes in equilibrium structure, which were outlined with the sequences of equilibrium. This effect is twofold: first, a higher degree of risk selection implies that the slope of the reinsurer's expected profits is reduced, and, second, the change in the portfolio of the reinsurer can imply drops in expected profits.

This is illustrated in Figure 4 displaying the expected profits of the reinsurer as a function of the premium rate for the three sequences of equilibria. It should be noted that the only

<sup>16</sup>Note that 
$$\frac{\partial \left(\frac{N_{N-N_G}}{N_S+N_G}\right)}{\partial \theta}$$
 and  $\mathcal{H}(\theta(1-\gamma)-p_H) < 0$  and hence the positive derivative for the  $HEH(r_S=0;r_G=1)$ .

<sup>17</sup>We have  $\frac{\partial \left(\frac{N_G}{N_S+N_G}\right)}{\partial \theta} < 0$  and thus  $\frac{\partial \pi_R}{\partial \theta}\Big|_{GE(r_G=0)} > \frac{\partial \pi_R}{\partial \theta}\Big|_{HEL(r_S=1;r_G=0)}$ . Further, manipulating the expression for the  $HEL(r_S=1;r_G=0)$  we can show that  $\frac{\partial \pi_R}{\partial \theta}\Big|_{HEL(r_S=1;r_G=0)} > \mathcal{H}(1-\gamma)$ .

difference between the sequences displayed is the difference between the damage probability in the H and L-region. In particular, overall expected damages ( $\mathcal{H}\overline{p}$ ) are identical for all sequences. In sequence i) (the change of retention regime within a generalist equilibrium) the transition from one type of equilibrium to another is smooth and occurs at zero expected profits for the reinsurer.

Figure 4 Evolution of expected profits of reinsurer



Parameter values: H = 100,000;  $\delta = 0.2$ ; f = 200; F = 200;  $\overline{p} = 0.05$ .  $p_H = 0.0525$  for sequence i),  $p_H = 0.065$  for sequence ii) and  $p_H = 0.08$  for sequence iii).

For sequence *ii*), the passage from the  $GE(r_G = 0)$  to the  $HEL(r_S = 1; r_G = 0)$  implies a change of slope (point I). Further, at point II, the passage from the  $HEL(r_S = 1; r_G = 0)$ to the  $SE(r_L = 1; r_H = 0)$ , expected profits drop first and then increase at an even lower rate. Finally, for sequence *iii*) there exists an early drop and change in slope of expected profits, when changing from the  $GE(r_G = 0)$  to a  $SE(r_L = 1; r_H = 0)$  (point III).

The graph illustrates that the tendency to risk selection can have important effects for the situation of the reinsurance company. For example, given our parameter values, for a premium rate of 0.06, corresponding to a situation where the reinsurer almost breaks even, expected losses are more than four times larger with partial risk selection (sequence ii)), and eight times larger with complete risk selection (sequence iii)).

## 6 Discussion

Within the more general discussion we want to address the following issues: 1) the evolution of the rate of reinsurance and profits in France; 2) the "social" cost of increasing the premium rate; 3) regions of different size; 4) risk-averse insurance companies; 5) introduction of a risk-pool transfer scheme as observed for some health insurance systems.

#### 6.1 The evolution of the rate of reinsurance and profits in France

Our model, based on the institutional setup in France, predicts that an increase in the premium rate is likely to lead to an increased degree of risk selection, which in turn, generally, reduces the average rate of reinsurance. Table 5 illustrates this result.<sup>18</sup> It displays the evolution of the rate of reinsurance from the beginning of the system in 1982/83 to 1997.<sup>19</sup> This period includes an initial raise in the premium rate occurring after 1984. In the years after this first premium increase, the rate of reinsurance decreased from close to 100% to 40%. Obviously, this could simply be an indication that the premium rate is on average "comfortably" high, such that the insurance companies need not reinsure fully. Unfortunately, data on the amount of risk selection is not available. Von Ungern provides some factual evidence stating that "(i)n 1995, the share of premiums ceded to the CCR was 45 percent. In the same year, the CCR had to pay 98 per cent of the damage caused by the three hurricanes in the DOM. With a premium volume of F24m (from the DOM), the CCR had to pay claims in the order of F650m..." (von Ungern (2004), p.91).

	Rate of	Accumulated	Premium income
Year	reinsurance	system's profit	of the CCR
	%	billion FF	billion FF
82/83	83	-1.4	1.6
84	75	0.5	1.9
85	75	3.2	2.2
86	73	5.9	2.2
87	52	8.2	1.8
88	41	10.1	1.5
89	51	13.3	1.5
90	43	14.0	1.5
91	40	17.4	1.5
92	38	19.7	1.5
93	41	20.8	1.7
94	41	24.4	1.9
95	45	26.5	2.1
96	39	29.0	2.0
97	40	33.5	2.1

 Table 5

 Evolution of the rate of reinsurance, profits of the system and premium income of the CCR

Source: von Ungern (2004).

The model also predicts that the reinsurance company cannot benefit from the profits generated in the system. This is also illustrated in Table 5 representing an estimation of the aggregate profits of the system as well as the evolution of revenues of the CCR. In the period 1982-1997 the system (and mainly the private insurers) accumulated an estimated 33,5 billion French Francs. At the same time, premium income of the CCR remained fairly stable over time, leading to the already mentioned need of refinancing of the reinsurer.

### 6.2 The "social" cost of increasing the premium rate

Our model allows us to calculate the "social" cost of the system. This cost is the sum of the expected losses of the reinsurer and the premium payments of the final consumers.<sup>20</sup>

<sup>&</sup>lt;sup>18</sup>Data in this Table are drawn from von Ungern (2004).

<sup>&</sup>lt;sup>19</sup>Recall that starting in 1999, the CCR only allowed a maximum of 50% reinsurance.

 $<sup>^{20}</sup>$ Recall that profits of the insurance companies are zero by the determination of equilibrium. Further, this is not a measure of "net welfare" as the (exogenous) cost of damage is not considered.

Note that, given the unique premium rate, the direct cost of insurance to final customers is simply  $2H\theta$ , i.e. the number of houses in the country times the premium rate.

Figure 5 illustrates the social cost as a function of the premium rate for the three equilibrium sequences defined. First of all, the figure illustrates that improving the situation of the reinsurance company by increasing the premium rate comes at an important cost to the taxpayer (and final consumer). This effect is accentuated (higher slope of the social cost curve) with more risk selection. Finally, we observe that bigger differences in the damage probabilities, sequences ii) and iii), might imply considerably higher social costs.



Figure 5 Social cost of the system

Parameter values: H = 100,000;  $\delta = 0.2$ ; f = 200; F = 200;  $\overline{p} = 0.05$ .  $p_H = 0.0525$  for sequence i),  $p_H = 0.065$  for sequence ii) and  $p_H = 0.08$  for sequence iii).

One should not overemphasize this simple welfare calculation. For one thing we do not include an analysis of why this institutional framework was chosen. Presumably, there is some benefit from the fact that a public reinsurance company exists, even if it generates losses. Our positive framework does not allow us to address the issue of the optimal institutional setup. For this, one would have to build a more complex model, which takes into account the preferences of the consumers.

However, we can analyze a situation where the private insurance companies are obliged to fully reinsure, and hence act as pure intermediaries. Figure 5 includes the social cost of such a system (the dashed-dotted line). Clearly, this simple change in the setup implies an important reduction of the total cost of the system. We can distinguish two effects that differentiate a policy of full reinsurance from the current institutional setup. First, when risk selection occurs, the social cost of the system increases much more strongly than under full reinsurance. Second, at a certain point the reinsurer actually makes positive profits and thus dampens the effect of higher premium rates.

#### 6.3 Different sized regions

Next we want to include different sized regions. It is easy to introduce different sizes of regions (see Appendix C), where it is realistic to assume that the *L*-region is bigger than the *H*-region.<sup>21</sup>

One might think that a bigger *L*-region would attenuate the problems outlined in our model. Thus, since an *a priori* overlook of the distribution of low and high damage probability areas in France would lead to the impression that the *L*-region is much bigger, one would be tempted to dismiss our results as hinging crucially on the identical size of the regions.

This is not the case. In fact, a bigger L-region might make things worse for the reinsurance company. Especially under a constant average damage probability, the consequences from risk selection are even more devastating, since now the reinsurance company must support a high fraction of total damages while only reinsuring a small part of the country housing stock.

Appendix C shows that the tendencies to risk selection persist virtually unchanged under different size regions. In a similar vein, Figure 6 below represents the total social cost of the system under a bigger *L*-region. For the example, we assume that the *L*-region is 10 times the size of the *H*-region and take slightly different values for  $p_H$ . The other parameters, in particular  $\overline{p}$  remain unchanged. Sequences a) and b) in the figure are similar to sequences ii) and iii) above. See Appendix C for details.



Figure 6 Social cost of the system - different size regions

Parameter values:  $M_L = 1,000,000; M_H = 100,000; \delta = 0.2; f = 200; F = 200; \overline{p} = 0.05.$  $p_H = 0.06$  for sequence a),  $p_H = 0.15$  for sequence b).

<sup>&</sup>lt;sup>21</sup>For clarity there is a slight change in notation in Appendix A3. The housing stock in region *i* is denoted by  $M_i$  (for maison).

The illustration shows that the problems of risk selection and their consequences on the cost of the system are present also under different size regions. In order to illustrate the additional costs causes by the risk selection we again included the cost of the system under the assumption that reinsurance is compulsory.

#### 6.4 Risk averse insurance companies

In this section we sketch a modification of the model to introduce risk-averse insurers. This implies interior solutions for the choice of the rate of retention. However, even though we introduce risk aversion in the simplest possible way, the model becomes intractable analytically, particularly for hybrid equilibrium situations.

We return to the initial model with equally sized regions. The setup is identical to the one outlined before, with the modification that insurers maximize expected utility

$$E\left[u\left(\pi_{j}\right)\right] = E\left[\pi_{j}\right] - \frac{\lambda}{2} Var\left[\pi_{j}\right]$$

where  $\lambda$  is the risk-aversion parameter,  $E[\pi_i]$  is given by (1) or (2) and

$$Var\left[\pi_{j}\right] = r_{j}^{2} \frac{\mathcal{H}}{N_{i}} p_{i} \left(1 - p_{i}\right)$$

for specialists and

$$Var\left[\pi_{j}\right] = r_{j}^{2} \left[\frac{\mathcal{H}}{N_{L}^{*}} p_{L} \left(1 - p_{L}\right) + \frac{\mathcal{H}}{N_{H}^{*}} p_{H} \left(1 - p_{H}\right)\right]$$

for generalists, choosing their rate of retention  $r_j$ . Before describing the equilibrium candidates, it is useful to introduce some further notation. We denote by  $\omega_{ij}$  the "risk-adjusted unitary profit" for insurer j in region i

$$\omega_{ij} = \left[\theta\gamma + r_j \left[\theta \left(1 - \gamma\right) - p_i\right] - \frac{\lambda}{2} \left(r_j\right)^2 p_i \left(1 - p_i\right)\right].$$

Equilibrium candidates are displayed in Table 6.<sup>22</sup>

 Table 6

 Risk-averse insurers: Equilibrium candidates

Eqm. candidate	Type of insurer	Rate of retention	Number of insurers
SE		$r_j^* = rac{ heta(1-\gamma)-p_i}{\lambda p_i(1-p_i)}$	$N_i = \frac{\mathcal{H}\omega_i}{f+F}$
GE		$r_G^* = \frac{\theta(1-\gamma) - p_L + \theta(1-\gamma) - p_H}{\lambda[p_L(1-p_L) + p_H(1-p_H)]}$	$N_G = \frac{\mathcal{H}\omega_G}{2f+F}$
HEL	Specialist	$r_S^* = rac{ heta(1-\gamma)-p_L}{\lambda p_L(1-p_L)}$	n.a
HEL	Generalist	$r_{G}^{*} = \frac{N_{G}[\theta(1-\gamma)-p_{L}] + N_{L}^{*}(\theta(1-\gamma)-p_{H})}{\lambda \left[ N_{G}p_{L}(1-p_{L}) + N_{L}^{*}p_{H}(1-p_{H}) \right]}$	n.a

Note: The displayed expression for the number of insurers corresponds to an interior solution for the rate of retention.

As for the main model, the next step is to derive conditions under which a given candidate is indeed an equilibrium. Since the candidate characterization for the hybrid equilibrium is not tractable, it is not possible to obtain analytical conditions in this case. One would have to proceed to a simulation exercise. For this reason we restrict the analysis to specialist

 $<sup>^{22}</sup>$ The number of insurers in the HEL specification is not a tractable expression, and thus is omitted in the table. For brevity we did not consider other types of hybrid equilibria.

and generalist equilibria. The equilibrium conditions are displayed in Table 7, recall the deviation strategy (D) for a specialist candidate is to offer in both regions, while it is to retreat to the *L*-region for the generalist candidate.

Equilibrium	Eqm. condition
SE	$ \begin{aligned} & f \left[ \frac{\mathcal{H}\omega_{LD}}{\mathcal{H}\omega_{L} + f + F} + \frac{\omega_{HD}}{\omega_{H}} - 2 \right] + \\ & F \left[ \frac{\mathcal{H}\omega_{LD}}{\mathcal{H}\omega_{L} + f + F} + \frac{\omega_{HD}}{\omega_{H}} - 1 \right] < 0 \end{aligned} $
GE	$\frac{\omega_L}{\omega_G} \left( 2f + F \right) - f - F < 0$

 Table 7

 Risk-averse insurers: Equilibrium conditions

As for the model with linear utility we observe that some conditions must be satisfied for a specific candidate to be an equilibrium. Thus, as before, a discrete increase in the premium rate might imply a change in the equilibrium configuration and lead to an increase in risk-selection, with the already explained consequences for the financial situation of the reinsurer. Thus the conclusions of this more general model would be essentially the same as for the model with risk-neutral insurers. The main difference in the results for the reinsurer is that its problems are somewhat attenuated, since for sufficient risk-aversion expected profit of the reinsurer might actually be positive.

#### 6.5 Risk-pool transfer scheme

So far we have considered one simple "mechanism" to improve the financial situation of the reinsurance company via the introduction of compulsory reinsurance. We have shown that such a scheme would significantly reduce the social cost of the system. Below we want to sketch an alternative; the introduction of a risk pool similar to the ones often encountered in health insurance. Generally, a risk pool is a mechanism to redistribute premium income from insurers with a good risk portfolio to insurers with a bad one.

We model the risk-pool as a payment (rp) for each good risk (i.e. insurance contract in the *L*-region). The payment rp is transferred to the reinsurance company, and is set such that the latter brakes even.

Thus, expected profits in the *L*-region are given by

$$E\left[\pi_{j}\right] = \frac{\mathcal{H}}{N_{L}^{*}}\left[\theta\gamma - rp + r_{j}\left(\theta\left(1-\gamma\right) - p_{i}\right)\right] - f - F,$$

for a specialist, and

$$E\left[\pi_{j}\right] = \frac{\mathcal{H}}{N_{L}^{*}}\left[\theta\gamma - rp + r_{j}\left(\theta\left(1-\gamma\right) - p_{L}\right)\right] + \frac{\mathcal{H}}{N_{H}^{*}}\left[\theta\gamma + r_{j}\left(\theta\left(1-\gamma\right) - p_{H}\right)\right] - 2f - F,$$

for a generalist. Note that expected profit for a specialist in the *H*-region are still given by (1), and the choice of the rate of retention for any insurer by (3) and (4). Similarly, expected profits of the reinsurer for the different equilibrium candidates are still given by the expressions in Table 3 adding  $\mathcal{H}rp$  to each expression.

Equilibrium candidates are now characterized by a system of equations considering the zero-profit conditions for each type of insurer in the candidate and, additionally, by the zero expected profit condition for the reinsurer to determine rp.

Descriptions of the equilibrium candidates are displayed in Table  $8.^{23}$ 

 $<sup>^{23}</sup>$ Since the risk-pool does not affect the *H*-region, the expressions for the number of insurers in this region are identical to the ones in Table 1.

Equilibrium	Number of ins. in <i>L</i> -region	Number of generalists
$SE(r_L = 0; r_H = 0)$	$\frac{\mathcal{H}[\theta - p_L + \theta(1 - \gamma) - p_H]}{(f + F)}$	
$SE(r_L = 1; r_H = 0)$	$\frac{\mathcal{H}[\theta - p_L + \theta(1 - \gamma) - p_L]}{(f + F)}$	
$GE\left(r_G=0\right)$		$\frac{\mathcal{H}[\theta - p_L + \theta - p_H]}{(2f + F)}$

Table 8Equilibrium candidates with risk pool

Equilibrium candidates where there is no reinsurance are not displayed, since in this case rp = 0 and the expressions are as in Table 1. Further, the hybrid equilibrium expressions are messy and are thus not displayed either. We observe similar expressions for the number of insurers as in our basic model. The main difference is that, in general, the number of insurers operating in the *L*-region is lower, due to the additional financing of the reinsurer.

From here the analysis is essentially the same as for the basic model, i.e. one could again define areas of equilibrium with the consequence that an discrete increase in the premium rate may lead to a change in the equilibrium configuration with an increase in risk-selection. All this is not needed to consider the social cost of such a system, since by construction expected profits of both the insurers and the reinsurer are zero. Hence, social cost is now determined uniquely by the cost to final customers. However, it should be mentioned that the risk-pool scheme is only feasible for premium rates that satisfy

$$(\theta - p_L) + (\theta - p_H) > 0$$

since, without this condition the number of insurers would become negative.

Figure 7 takes up Figure 5 and includes the social cost of the risk-pool scheme. Quite intuitively, the slope of the risk-pool scheme is the highest and coincides with the segments of the above mentioned sequences for situations where there is no reinsurance. In this sense, the risk pool has (weakly) lower costs than the current situation of free choice of reinsurance, whenever the risk-pool is feasible. Further, we can also detect a range of premium rate (in our case between 0.05 and slightly more than 0.06), where the risk-pool scheme is also better than the alternative of full reinsurance.

## 7 Conclusions

Natural disaster insurance has recently come back into the policy agenda due to increased occurrence and severity of floods and other catastrophes. The situation in France merits particular attention because of some stylized facts one can observe. Although the system is financially viable (the claims/premium ratio since introduction of the insurance is around 55%), the existing, publicly owned, reinsurance company (Caisse Centrale de Réassurance - CCR) was never able to build up vital reserves and had to be recapitalized in 1999. This injection of public funds was accompanied by an increase in premium rates and excesses of 30%.

Our claim is that this situation is the outcome of flaws in the institutional setup. The private insurance companies are free to contract reinsurance at any degree with the CCR. This leads to a situation of risk selection by the insurance companies. Further, the increase in the premium rates to improve the situation for the financially distressed CCR, comes at a high cost for the tax payer and final consumer of the insurance.

Figure 7 Social cost - risk pool



Parameter values same as in Figure 5.

In order to sustain this claim we build a model that represents the situation of the French property insurance market. Although the model is kept as simple as possible, we include the main features of the institutional setup, such as the uniform premium rate across the country and the existence of a (publicly owned) reinsurer that offers (proportional) reinsurance cover. The equilibrium outcomes of our model confirms the inherent tendency of risk selection. In particular, assuming redistributive premium rates, where low damage probability regions subsidize high damage probability regions, no equilibrium where all insurance companies serve the whole country ("generalist equilibrium") can exist. Further, we find that a higher premium rate is likely to increase the tendency for risk selection. Note that the increase in risk selection is not gradual. Rather, a sufficient increase in the premium rate eventually leads to a change in the equilibrium market configuration, one with a higher degree of risk selection.

These changes in equilibrium configuration have their effect on the reinsurer. While a marginal increase in the premium rate improves the expected profits of the reinsurer, this effect is dampened with a higher degree of risk selection. When calculating the "social" cost of the system, which is defined as the sum of (final) consumer premium payments and expected profits (losses) of the reinsurer, we show that the current institutional setup leads to high social cost. In particular, social costs of the current setup are (in general) significantly higher than in a situation where all insurers are obliged to fully reinsure and act as pure intermediaries.

## References

- Jaffee, Dwight M. and Thomas Russell (1996): Catastrophe Insurance, Capital Markets and Uninsurable Risks, The Wharton School Financial Institutions Center 96-12, University of Pennsylvania.
- [2] Mas-Colell, Andreu, Michael D. Whinston and Jerry Green (1995): Microeconomic Theory, Oxford University Press, New York, London.
- [3] Michel-Kerjan, Erwann (2001): Insurance against Natural Disasters: Do the French Have the Answer? Strengths and Limitations, Cahier n°2001-007, Ecole Polytechnique.
- [4] von Ungern Sternberg, Thomas (2004): Efficient Monopolies The Limits of Competition in the European Property Insurance Market, Oxford University Press.

## A Determination of equilibrium candidates

Below we describe in some more detail the determination of the equilibrium candidates for each of the equilibrium defined in the text.

#### Specialist equilibrium candidates

Due to the discrete optimal rate of retention we can distinguish three characterizations of a potential specialist equilibrium:

Characterization	$\theta \left( 1 - \gamma \right) - p_L$	$\theta \left( 1 - \gamma \right) - p_H$
$SE(r_L = 0; r_H = 0)$	<0	<0
$SE(r_L = 1; r_H = 0)$	>0	<0
$SE(r_L = 1; r_H = 1)$	>0	>0

In each case the equilibrium number of firms will be determined by setting (1) equal to zero and solving for the equilibrium number of firms. Denoting by  $N_i^S$  the number of firms in region *i* in the specialist equilibrium characterization we obtain:

Eqm. Candidate	$N_L^S$	$N_H^S$
$SE(r_L = 0; r_H = 0)$	$\frac{\mathcal{H}\theta\gamma}{(f+F)}$	$\frac{\mathcal{H}\theta\gamma}{(f+F)}$
$SE(r_L = 1; r_H = 0)$	$\frac{\mathcal{H}(\theta - p_L)}{(f + F)}$	$\frac{\mathcal{H}\theta\gamma}{(f+F)}$
$SE(r_L = 1; r_H = 1)$	$\frac{\mathcal{H}(\theta - p_L)}{(f + F)}$	$\frac{\mathcal{H}(\theta - p_H)}{(f + F)}$

#### Generalist equilibrium candidates

For the generalist equilibrium candidates there are only two characterizations according to (4). The equilibrium number of firms, denoted by  $N_G$ , is determined by setting (2) equal to zero:

Eqm. Candidate	$N_G$
$GE\left(r_G=0\right)$	$\frac{2\mathcal{H}\theta\gamma}{(2f+F)}$
$GE\left(r_G=1\right)$	$\frac{\mathcal{H}(\theta - p_L + \theta - p_H)}{(2f + F)}$

#### Hybrid equilibrium characterization

In a so called hybrid equilibrium candidate (HE) the choice of the rate of retention will again be determined by (3) and (4), respectively.<sup>24</sup>

The equilibrium number of insurers is obtained by a zero-profit condition for each type of firm, using (1) and (2). Note that we must distinguish between HEL and HEH candidates. For example, in a HEH characterization the conditions to determine the number of each type of insurer are:

$$E [\pi_S] = \frac{\mathcal{H}}{N_H^*} [\theta \gamma + r_S (\theta (1 - \gamma) - p_H)] - f - F = 0$$
  

$$E [\pi_G] = \frac{\mathcal{H}}{N_G} [\theta \gamma + r_G (\theta (1 - \gamma) - p_L)]$$
  

$$+ \frac{\mathcal{H}}{N_H^*} [\theta \gamma + r_G (\theta (1 - \gamma) - p_H)] - 2f - F = 0$$

where  $\pi_S$  and  $\pi_G$  are profits of the specialists and generalists, respectively, and  $N_H^* = N_S + N_G$  is the total number of insurers operating in the *H*-region (and also the total number of firms in the country). Again, we must consider the four cases of possible rates of retention.

Below we present the equilibrium number of insurers for the hybrid equilibrium candidates when the specialists operate in the *L*-region and the *H*-region. We indicate the total as well as the generalist number of insurers. The number of specialists can easily be obtained by difference. Note that the table only excludes situations, denoted by (-), where the number of one type of firms is negative under all parameter constellations.<sup>25</sup>

Eqm. Candidate	$N_L^*$	$N_G^{HEL}$
$HEL\left(r_S=0;r_G=0\right)$	_	—
$HEL\left(r_S=0;r_G=1\right)$	$rac{\mathcal{H} heta\gamma}{(f+F)}$	$\frac{\mathcal{H}(\theta - p_H)}{f\left(2 - \frac{\theta - p_L}{\theta\gamma}\right) + F\left(1 - \frac{\theta - p_L}{\theta\gamma}\right)}$
$HEL\left(r_S=1;r_G=0\right)$	$\frac{\mathcal{H}(\theta - p_L)}{(f + F)}$	$\frac{\mathcal{H}\theta\gamma}{f\left(2-\frac{\theta\gamma}{\theta-p_L}\right)+F\left(1-\frac{\theta\gamma}{\theta-p_L}\right)}$
$HEL\left(r_S=1;r_G=1\right)$	$\frac{\mathcal{H}(\theta - p_L)}{(f + F)}$	$rac{\mathcal{H}(\theta - p_H)}{f}$

Eqm. Candidate	$N_H^*$	$N_G^{HEH}$
$HEH\left(r_S=0;r_G=0\right)$	_	_
$HEH\left(r_S=0;r_G=1\right)$	$rac{\mathcal{H} heta\gamma}{(f+F)}$	$\frac{\mathcal{H}(\theta - p_L)}{f\left(2 - \frac{\theta - p_H}{\theta\gamma}\right) + F\left(1 - \frac{\theta - p_H}{\theta\gamma}\right)}$
$HEH\left(r_S=1;r_G=0\right)$	$\frac{\mathcal{H}(\theta - p_H)}{(f + F)}$	$\frac{\mathcal{H}\theta\gamma}{f\left(2-\frac{\theta\gamma}{\theta-p_{H}}\right)+F\left(1-\frac{\theta\gamma}{\theta-p_{H}}\right)}$
$HEH\left(r_S=1;r_G=1\right)$	_	_

We observe that a HE candidate with a rate of retention of zero is not possible. In fact, under such a situation the number of specialists would be negative. This result is quite intuitive, under zero retention an insurer which operates in both regions has an advantage over an insurer specializing in one region, since it can "dilute" the country fixed costs. Note further, that within the HEH ( $r_S = 1; r_G = 1$ ) candidate there is additionally no possibility of a situation with full retention. Again in this case the number of specialist firms would be negative.

 $<sup>^{24}</sup>$ One could imagine a hybrid equilibrium with three type of firms: specialist in either region and generalists. However, for the characterization of such an equilibrium a condition, which is independent of the number of firms of any type, must hold. We do not consider this particular case.

 $<sup>^{25}</sup>$ We therefore, so far, do not rule out potential negative numbers of firms under certain parameter values. We will introduce this restriction later on in the analysis.

Next we must check whether the potential equilibrium characterizations are feasible in the sense that there exist situations where the rate of retention of the equilibrium characterization indeed is an optimal choice for the reinsurer. If this is not the case, we can eliminate further candidates.

For example, within the  $HEL(r_S = 0; r_G = 1)$  characterization we have that  $r_S = 0$ . This implies for an optimal choice, from (3), that  $\theta(1 - \gamma) - p_L < 0$ . Hence, under  $p_H > p_L$  we also have  $\theta(1 - \gamma) - p_H < 0$ , which implies using (4), that  $r_G = 1$ , is never an optimal choice. We are thus able to exclude the  $HEL(r_S = 1; r_G = 1)$  candidate. For the other two cases there are no restrictions that would render them a priori infeasible.

An inverse reasoning shows that the  $HEH(r_S = 1; r_G = 0)$  candidate is not feasible. Indeed,  $r_S = 1$  implies  $\theta(1 - \gamma) - p_H > 0$ , which implies that the optimal choice for a generalist is  $r_G = 1$ . Thus we can also eliminate this characterization.

Eqm. Candidate	$N_i^*$	$N_G^{HEi}$
$HEL\left(r_S=1;r_G=0\right)$	$\frac{\mathcal{H}(\theta - p_L)}{(f + F)}$	$\frac{\mathcal{H}\theta\gamma}{f\left(2-\frac{\theta\gamma}{\theta-p_L}\right)+F\left(1-\frac{\theta\gamma}{\theta-p_L}\right)}$
$HEL(r_S=1;r_G=1)$	$\frac{\mathcal{H}(\theta - p_L)}{(f + F)}$	$rac{\mathcal{H}( heta-p_H)}{f}$
$HEH\left(r_S=0;r_G=1\right)$	$\frac{\mathcal{H}\theta\gamma}{(f+F)}$	$\frac{\mathcal{H}(\theta - p_L)}{f\left(2 - \frac{\theta - p_H}{\theta\gamma}\right) + F\left(1 - \frac{\theta - p_H}{\theta\gamma}\right)}$

Summarizing, the potential hybrid equilibrium candidates are:

#### Comparative statics of the equilibrium characterizations

Table A-1 below summarizes the comparative statics effects of the parameters in the model on the number of insurers in each equilibrium candidate<sup>26</sup>. Results are in general straightforward and intuitive. The only somewhat different effects occur in the hybrid equilibrium candidates. There we observe that some derivatives cannot be signed, or have the opposite sign as the ones observed for the specialist and generalist candidates. This is due to the fact that one type of insurer can "benefit" from an increase in one of the parameters at the cost of the other type in the equilibrium. Generally, the pattern is the following: the number of houses ( $\mathcal{H}$ ), the premium rate ( $\theta$ ) as well as the commission rate ( $\gamma$ ) have a positive effect on the number of insurers, while the damage probabilities ( $p_i$ ) and fixed costs have a negative effect. As exceptions to these effects we can note that the damage probability in the *L*-region affects positively the number of generalist firms in a HEL ( $r_S = 1, r_G = 0$ ) situation. Whereas for the same candidate the rate of commissions ( $\gamma$ ) affects negatively the number of specialists. The effect of an increase in the premium rate for this candidate is ambiguous. Further, note that the damage probabilities have a positive effect on the number of specialists in the HEH ( $r_S = 0; r_G = 1$ ) candidate.

<sup>&</sup>lt;sup>26</sup>Note that the effect of an increase in the premium rate ( $\theta$ ) is further analyzed in a subsequent section.

Eqm. candidate	Type of firm	н	θ	$p_L$	$p_H$	$\gamma$	f, (F)
$SE(r_L = 0; r_H = 0)$	$N_L$	>	>	0	0	>	<
	$N_H$	>	>	0	0	>	<
$SE(r_L = 1; r_H = 0)$	$N_L$	>	>	<	0	0	<
	$N_H$	>	>	0	0	>	<
$SE(r_L = 1; r_H = 1)$	$N_L$	>	>	<	0	0	<
	$N_H$	>	>	0	<	0	<
$GE\left(r_{G}=0\right)$	$N_G$	>	>	0	0	>	<
$GE(r_G=1)$	$N_G$	>	>	<	<	0	<
$HEL(r_S=1; r_G=0)$	$N_G$	>	≶	>	0	>	<
	$N_S$	>	≶	<	0	<	≶
$HEL(r_S=1; r_G=1)$	$N_G$	>	>	0	<	0	< (0)
	$N_S$	≶	<	<	>	0	$\leq (<)$
$HEH(r_S=0;r_G=1)$	$N_G$	>	≶	<	<	$\leq$	<
	$N_S$	≶	≶	>	>	$\leq$	≶

 Table A-1

 Comparative Statics of equilibrium candidates

## **B** Equilibrium conditions

We consider the deviation strategies for the three types of equilibrium: specialist, generalist and hybrid.

#### Specialist equilibrium

Under the specialist equilibrium candidates we have separation of markets. We must consider deviation strategies for the H and the L regions. However, the possible deviations are of the same kind in both cases. Does an insurer have an incentive to offer in both regions, choosing its optimal rate of retention given by (4)? It seems intuitive that an insurer operating in the H-region more likely has an incentive to deviate. For this reason we concentrate on the situation for a specialist firm operating in the high probability region. Below we present the corresponding conditions for a specialist in the L-region. Note that the deviation implies that the number of market participants does not change in the region where the insurer was already active, while it increases by one in the other region. For deviation from the H-region this implies:

$$N_H = N_H^S$$
$$N_L = N_L^S + 1$$

Further we can distinguish four possible cases that potentially occur, characterized in the table below. We denote by  $r_D$  the optimal choice of the rate of retention for a deviation strategy.

Char.	Case	$r_D$
$SE(r_L = 0; r_H = 0)$		0
$SE(r_L = 1; r_H = 0)$	1	0
$SE(r_L = 1; r_H = 0)$	2	1
$SE(r_L = 1; r_H = 1)$		1

To obtain the condition under which a deviation is not profitable we compute the expected profits of a deviating insurer, taking into account the number of firms in the market. This expression must be negative, implying the conditions we are looking for.

$$SE(r_L = 0; r_H = 0)$$

Deviating expected profits are given by:

$$E\left[\pi_{D}\right] = \frac{\mathcal{H}}{N_{L}}\theta\gamma + \frac{\mathcal{H}}{N_{H}}\theta\gamma - 2f - F < 0$$

Using the expression for the number of insurers we obtain:

$$E\left[\pi_{D}\right] = \frac{\mathcal{H}\theta\gamma}{\frac{\mathcal{H}\theta\gamma+f+F}{f+F}} + \frac{\mathcal{H}\theta\gamma}{\frac{\mathcal{H}\theta\gamma}{f+F}} - 2f - F < 0 \qquad (B-1)$$

$$= \frac{\mathcal{H}\theta\gamma}{\mathcal{H}\theta\gamma + f + F} \left(f + F\right) - f < 0 \tag{5}$$

If this condition holds, then a typical insurer of a  $SE(r_L = 0; r_H = 0)$  operating in the H-region, will not have an incentive to deviate from the equilibrium characterization<sup>27</sup>.

### $\mathbf{SE}(\mathbf{r}_L \!=\! \mathbf{1}; \mathbf{r}_H \!=\! \mathbf{0})$ Case 1

The same procedure leads to the following condition:

$$E[\pi_D] = \frac{\mathcal{H}\theta\gamma}{\mathcal{H}(\theta - p_L) + f + F} (f + F) - f < 0 \qquad (B - 2)$$

The condition states that deviation is not profitable if generated income from entering the *L*-region does not cover the additional regional fixed costs.

### $\mathbf{SE}(\mathbf{r}_L = \mathbf{1}; \mathbf{r}_H = \mathbf{0})$ Case 2

The condition for this case is somewhat more complicated. Again, determining the expected profits from deviation, using the equilibrium number of firms and then manipulating the expression somewhat, we obtain:

$$E[\pi_D] = \begin{cases} f\left[\frac{\mathcal{H}(\theta - p_L)}{\mathcal{H}(\theta - p_L) + f + F} + \frac{(\theta - p_H)}{\theta\gamma} - 2\right] \\ +F\left[\frac{\mathcal{H}(\theta - p_L)}{\mathcal{H}(\theta - p_L) + f + F} + \frac{(\theta - p_H)}{\theta\gamma} - 1\right] \end{cases} < 0 \tag{B-3}$$

Note that the term in f is strictly negative. This comes from the fact that here (3) in the *H*-region is negative and thus:  $\theta - p_H < \theta \gamma$ . On the other hand, the term in F is potentially positive.

#### $SE(r_L = 1; r_H = 1)$

Finally the condition a in  $SE(r_L = 1; r_H = 1)$  is given by:

$$E[\pi_D] = \frac{\mathcal{H}(\theta - p_L)}{\mathcal{H}(\theta - p_L) + f + F} (f + F) - f < 0 \qquad (B - 4)$$

The corresponding conditions for a specialist in the L-region are the following:

 $<sup>^{27}</sup>$ In this case, as is shown in the appendix, the condition for a *L*-insurer is the same, since both have the same initial rate of retention.

Candidate	Condition
$SE(r_L = 0; r_H = 0)$	$\frac{\mathcal{H}\theta\gamma}{\mathcal{H}\theta\gamma+f+F}\left(f+F\right)-f<0$
$SE(r_L = 1; r_H = 0)$ Case 1	$ \begin{aligned} & f \left[ \frac{\theta \gamma}{(\theta - p_L)} + \frac{\mathcal{H}\theta \gamma}{\mathcal{H}\theta \gamma + f + F} - 2 \right] \\ & + F \left[ \frac{\theta \gamma}{(\theta - p_L)} + \frac{\mathcal{H}\theta \gamma}{\mathcal{H}\theta \gamma + f + F} - 1 \right] \end{aligned} < 0 \end{aligned}$
$SE(r_L = 1; r_H = 0)$ Case 2	$\frac{\mathcal{H}(\theta - p_H)}{\mathcal{H}\theta\gamma + f + F} \left( f + F \right) - f < 0$
$SE(r_L = 1; r_H = 1)$	$\frac{\mathcal{H}(\theta - p_H)}{\mathcal{H}(\theta - p_H) + f + F} \left( f + F \right) - f < 0$

#### Generalist equilibrium

To determine the conditions under which deviation from a generalist equilibrium candidate is not profitable we proceed in the exact same way as for the specialist equilibrium characterization. Here the potential deviation strategy is to restrict service into one of the regions. However, we can show that deviation into the high probability region is never profitable.

Thus, the conditions we must check corresponds to deviation into the L-region, choosing the rate of retention according to (3). Note that restricting service into one region does not change the number of insurers active in that particular region. Here we can distinguish three cases.

Candidate	Case	$r_D$
$GE\left(r_G=0\right)$	1	0
$GE\left(r_G=0\right)$	2	1
$GE\left(r_G=1\right)$		1

#### $\mathbf{GE}\left(\mathbf{r_G}=\mathbf{0}\right)\ \mathbf{Case}\ \mathbf{1}$

Expected profits from deviation are given by:

$$E[\pi_D] = \frac{\mathcal{H}\theta\gamma}{N_G} - f - F$$
$$= \frac{\mathcal{H}\theta\gamma}{\frac{2\mathcal{H}\theta\gamma}{(2f+F)}} - f - F$$
$$= f + \frac{F}{2} - f - F < 0$$

Thus, deviation from a  $GE(r_G = 0)$  is never profitable. A situation which is quite intuitive. Here insurers fully reinsure and the zero profit condition is met with firms "diluting" the country fixed cost over two regions. Retiring from one region in such a situation, only reducing the regional fixed costs, is not profitable.

#### $\mathbf{GE}\left(\mathbf{r_G}=\mathbf{0}\right)\ \mathbf{Case}\ \mathbf{2}$

Calculating expected profits from deviating, using the equilibrium number of firms and then manipulating the expression we obtain the following condition:

$$E\left[\pi_D\right] = f\left[\frac{(\theta - p_L)}{\theta\gamma} - 1\right] + F\left[\frac{(\theta - p_L)}{2\theta\gamma} - 1\right] < 0 \qquad (B - 5)$$

Note that here the term in f is strictly positive, since  $r_D = 1$  implies  $(\theta - p_L) > \theta \gamma$ .

$$\mathbf{GE}\left(\mathbf{r_G}=\mathbf{1}\right)$$

Similarly, the condition for  $GE(r_G = 1)$  is given by:

$$E\left[\pi_{D}\right] = f\left[\frac{2\left(\theta - p_{L}\right)}{\left(\theta - p_{L} + \theta - p_{H}\right)} - 1\right] + F\left[\frac{\left(\theta - p_{L}\right)}{\left(\theta - p_{L} + \theta - p_{H}\right)} - 1\right] < 0$$

Note again that in this case the term in f is strictly positive. Further it is interesting to note that in the case of  $p_H > \theta$ , the condition is never satisfied, i.e. in a situation where there is redistribution between the low and the high probability region, a  $GE(r_G = 1)$  candidate is never an equilibrium. We will come back to this issue later on.

#### Hybrid equilibrium

#### HEL

For these candidates the only possible deviating strategy is to restrict service to the H-region. In fact, if a specialist would want to become a generalist, then the number of generalist insurers would increase by one, leading to negative expected profits due to the zero-profit condition imposed on the characterization. Similarly, if a generalist restricts service to the L-region the number of insurers in that region does not change. Hence, its situation becomes identical to the already active specialists in that region, which have zero expected profits. Finally, note that we can restrict the analysis to generalists. If one of them deviates, the number of insurers in the H-region does not change, whereas if a specialist deviates the number of firms in the market increases making the situation worse.

As we have seen before, there remain two possible candidates for this hybrid equilibrium:  $HEL(r_S = 1; r_G = 0)$  and  $HEL(r_S = 1; r_G = 1)$ .

In the first case, we can show that  $r_G = 0$  implies that  $\theta < p_H$ . Thus, the optimal rate of retention for the deviating insurer is  $r_D = 0$ . Expected profits become:

$$E\left[\pi_D\right] = \frac{\mathcal{H}\theta\gamma}{N_G} - f - F < 0$$

using the expression for  $N_G$  in this case we obtain:

$$E\left[\pi_{D}\right] = f - \frac{\theta\gamma}{\left(\theta - p_{L}\right)}\left(f + F\right) < 0$$

It should be recalled that  $r_S = 1$  implies  $(\theta - p_L) > \theta \gamma$ , and thus the condition is not necessarily negative.

For the  $HEL(r_S = 1; r_G = 1)$  we can show that deviation is never profitable.

#### HEH

Within the hybrid equilibrium candidate where specialists are active in the high probability region, only one potential case is actually possible. The condition for an equilibrium in this case can be found in the same way as before. The deviation strategy for a generalist is to restrict service to the *L*-region. The condition on the rate of retention for the generalist implies that  $(\theta - p_L) > \theta \gamma$  must hold. Thus the optimal rate of retention for deviation is  $r_D = 1$ . Expected profits from deviation can then be written as:

$$E\left[\pi_D\right] = \frac{\mathcal{H}\left(\theta - p_L\right)}{N_G} - f - F$$

which becomes, using the expression for the number of insurers:

$$E\left[\pi_D\right] = f - \frac{\left(\theta - p_H\right)}{\theta\gamma} \left(f + F\right) < 0$$

We can note here that deviation is always profitable in situations where  $\theta < p_H$ .

#### Minimum number of insurers

Finally, before moving to the next step, we want to restrict the set of equilibria such that at least one firm is active in the market. We do this by specifying conditions ensuring that within the most unfavorable situation there is still at least one firm in the market. For the specialist equilibrium this is given by:

$$N_{H}^{SE} = \frac{\mathcal{H}\theta\gamma}{(f+F)} \ge 1$$
$$\implies \mathcal{H}\theta\gamma \ge (f+F)$$

and for the generalist equilibrium:

$$N_G = \frac{2\mathcal{H}\theta\gamma}{(2f+F)} \ge 1$$
$$\implies \mathcal{H}\theta\gamma \ge \frac{(2f+F)}{2}$$

for the hybrid equilibrium the conditions that at least one firm of each type is active are not very handy. Thus we impose the restriction that the total number of insurers is at least two:

$$N_i^* = \frac{\mathcal{H}\theta\gamma}{(f+F)} \ge 2$$
$$\implies \mathcal{H}\theta\gamma \ge 2(f+F)$$

Note that, quite intuitively, the least restrictive condition is the one on the generalist equilibrium.

#### Description of the graphical analysis

#### Specialist equilibrium

We start with the specialist equilibrium and present only the lines corresponding to insurers operating in the H-region.<sup>28</sup>

Figure B-1 below presents the switch-lines for the  $SE(r_L = 1; r_H = 0)$  in the premium rate  $(\theta)$  - high risk area probability  $(p_H)$  space. The upper increasing line (rh) represents situations where an H-insurer wants to change retention regime. Above rh the premium rate is favorable enough that the insurer retains all risks  $(r_H = 1)$ , whereas below the line the insurer fully reinsures  $(r_H = 0)$ . Similarly, the lower increasing line (rd) represents the switch-line for the deviating insurer.<sup>29</sup> Again, above rd the deviating insurer chooses full retention. Finally, the decreasing line (rl) corresponds to the switch-line for an L-insurer, with full retention above the line. Thus the area for a  $SE(r_L = 1; r_H = 0)$  Case 1 is delimited above by rd and below by rl. Correspondingly,  $SE(r_L = 1; r_H = 0)$  Case 2 is delimited above by rh and below by rd.

 $<sup>^{28}</sup>$ We assume that these conditions are more stringent than the ones for insurers in the L-region. Thus we assume that, at the margin, if an H-insurer wants to proceed to a change an L-firm will not want to do so, yet.  $^{29}$ Recall that a deviating insurer from a specialist equilibrium characterization will necessarily choose to be

a generalist. Further, note that by considering H-region conditions, deviation implies that  $N_L = N_L^{SE} + 1$ .

Figure B-1 Switch-lines for  $SE(r_L = 1; r_H = 0)$  characterization



Parameter values:  $H = 100,000; \delta = 0.2; f = 200; F = 200; \overline{p} = 0.05.$ 

The relevant jump-lines for the  $SE(r_L = 1; r_H = 0)$  are presented in Figure B-2. The increasing line corresponds to the jump-line for  $SE(r_L = 1; r_H = 0)$  Case 1, i.e. situations where (B - 2) holds with equality. Below the line the equilibrium condition is satisfied. Similarly, the decreasing line represents situations where (B - 3) holds with equality. In other words, this is the jump-line for  $SE(r_L = 1; r_H = 0)$  Case 2. Here the condition holds everywhere above the line.

Figure B-3 takes the two graphs together. It allows to determine the area of equilibria for the particular candidate under study  $(SE(r_L = 1; r_H = 0))$ .<sup>30</sup> In our example this area is principally determined by the relevant jump-lines. However, the switch-lines are also important because they delimit the range where a specific jump line is actually relevant. In order to illustrate this interplay we have included in Figure B-3 the switch-line for the deviating firm (rd). In the end, the area of equilibria is determined by the area which is both below the dotted  $(J(r_L = 1, r_H = 0, r_D = 1))$  and above the solid line  $(J(r_L =$  $1, r_H = 0, r_D = 0))$ . To get to this conclusion we must consider also the switch-line for the deviating insurer (rd). Take for example the solid jump-line  $(J(r_L = 1, r_H = 0, r_D = 0))$ . This line, by definition is only valid for situation where the deviating insurer chooses a zero rate of retention, hence situation which are delimited from above by rd. Anywhere else this is not the relevant jump-line. Similar argumentation implies that another part of the area of equilibria is delimited by  $J(r_L = 1, r_H = 0, r_D = 1)$  from above and rd from below. Taking these parts together, we obtain the full (shaded) area of equilibria.

 $<sup>^{30}\</sup>mathrm{For}$  clarity the lines irrelevant for equilibrium determination are not drawn.



Parameter values:  $H = 100,000; \delta = 0.2; f = 200; F = 200; \overline{p} = 0.05.$ 

Figure B-3 Equilibrium Area for the  $SE(r_L = 1; r_H = 0)$ 



Parameter values:  $H = 100,000; \delta = 0.2; f = 200; F = 200; \overline{p} = 0.05.$ 

Finally, Figure B-4 includes the last restriction, the fact that at least one insurer should be active in every region, line denoted by NF.



Parameter values:  $H = 100,000; \delta = 0.2; f = 200; F = 200; \overline{p} = 0.05.$ 

Similar steps can be applied to obtain the equilibrium area for all the SE-configurations, which are shown in the first column of Figure B-5. Note that under our parameter specifications there are no equilibria of type  $SE(r_L = 0; r_H = 0)$  and  $SE(r_L = 1; r_H = 1)$ , due to the fact that some conditions are mutually exclusive. For the  $SE(r_L = 0; r_H = 0)$  it is the fact that the jump-line and the restriction for the minimum number of firms coincide, while for the  $SE(r_L = 1; r_H = 1)$  the switch and the jump-line delimit exclusive areas.

In the same figure we have a look at how the "shape" of the equilibrium area changes under different parameter constellations. Of particular interest are changes in the structure of the fixed costs. For this we specify two alternative scenarios, one where country fixed costs (F) are high and one where regional fixed costs (f) are high. In both cases we leave the sum of fixed costs (f + F) identical to the base scenario. Column two and three present the resulting area of equilibria.

For the  $SE(r_L = 0; r_H = 0)$  we have that equilibrium is delimited from above the by the jump-line  $J(r_L = 0, r_H = 0, r_D = 0)$ . As mentioned above, there is no equilibrium of this characterization under the base parameter values. Similarly, a high country fixed cost (F) does not give an equilibrium, whereas high regional fixed costs (f) imply existence of  $SE(r_L = 0; r_H = 0)$ -type equilibria. This result is quite intuitive, only in situations where regional fixed costs are high relative to country fixed costs does it make sense to specialize into one type of market in a situation of full reinsurance. In the other situations it is more appropriate to offer in both regions, in order to dilute the country fixed cost.

Regarding the  $SE(r_L = 1; r_H = 0)$  (in row two) we can observe that a change in the relative composition of fixed costs moves the relevant jump-lines around. The result is that a  $SE(r_L = 1; r_H = 0)$  equilibrium can be sustained only in case of extreme differences in damage probabilities in the case of high F, whereas the area of equilibrium is much bigger under high f. Again, this result is intuitive in that high country fixed cost provide an incentive for specialization only in the case of important differences across the regions.

Finally, under the considered parameter constellations there is no  $SE(r_L = 1; r_H = 1)$  equilibrium. Again we observe that changes in the fixed cost composition move around the jump-line.

#### Generalist Equilibrium

Having introduced the different concepts we use for the graphical analysis, we can proceed much quicker for the other two equilibrium characterizations. As before, within the Generalist Equilibrium candidates, equilibria are determined through the interplay between the relevant switch- and jump-lines. Figure B-6 presents these results for the two candidates of GE ( $GE(r_G = 0), GE(r_G = 1)$ ). The figure also includes the effects of a change in the relative composition of fixed costs.

Above we have shown that a  $GE(r_G = 0)$  with  $r_D = 0$  is always an equilibrium. Since, in our case  $r_D = r_L$ , we have that all points below the rl line are Generalist Equilibria, except for the restriction on the minimum number of insurers. Further, points below the jump-line for  $GE(r_G = 0)$  are also part of the area of equilibria. Finally, note that here an increase in the relative magnitude of F implies an increase in the area of equilibrium.

For the  $GE(r_G = 1)$  we have that situations with high premium rates and little differences in damage probabilities constitute an equilibrium. Again the area of equilibria expands when F is relatively high and contracts for high f.

#### Hybrid Equilibrium

Figure B-7 presents the area of equilibrium for the three HE candidates (HEL ( $r_S = 1; r_G = 0$ ), HEL ( $r_S = 1; r_G = 1$ ), HEH ( $r_S = 0; r_G = 1$ )) including the effects of a change in the relative composition of fixed costs.

We observe that a  $HEL(r_S = 1; r_G = 0)$  equilibrium is delimited from below by rl and from above by rg and the relevant jump-line. Note that, as in the GE case, a high F increases the area of equilibria.

As for the  $HEL(r_S = 1; r_G = 1)$ , we have shown above that it is always an equilibrium. Thus, all points above rg represent an equilibrium. Note that changes in the composition of fixed costs only slightly alter rg, and hence the area of equilibria.

Finally, within our parameter values we observe that the area of a HEH ( $r_S = 0; r_G = 1$ ) is rather small, again with its size increasing with relatively high country fixed costs.

Figure B-5 Specialist equilibrium



Parameter values: H = 100,000;  $\delta = 0.2$ ; f = 200; F = 200;  $\overline{p} = 0.05$  for base scenario. f = 100; F = 300 for "High F" and f = 300; F = 100 for "High f".

Figure B-6 Generalist equilibrium

Base

High F

High f

GE(0)



GE(1)



Parameter values: H = 100,000;  $\delta = 0.2$ ; f = 200; F = 200;  $\overline{p} = 0.05$  for base scenario. f = 100; F = 300 for "High F" and f = 300; F = 100 for "High f".

Base High F High f 0.14 0.14 0.12 0.14 0.12 0.12 0.1 0.1 0.08 HEL(1;0) theta theta 0.08 theta 0.00 0.06 -HEL(1;0) HED(1;0) 0.06 0.06 HEL(1:0) 0.04 0.04 0.04 0.02 0.02 0.02 0 0.05 0.06 0.07 0.08 0.09 0.1 0 0.06 0.06 0.07 0.06 6.00 0 0.05 0.06 0.07 0.06 0.09 \_\_\_\_\_\_ph\_jjd=1, rg=0, rd=0) 0.14 0.14 0.14 0.12 0.12 0.12 HEL(1;1) HEL(1;1) HEL(1;1) 0.1 0.1 0.1 theta 0.08 thata 0.08 HEL(1;1) 0.08 theta 0.06 0.06 0.06 0.04 0.04 0.04 0.02 0.02 0.02 0.09 0005 0.06 0.08 0.05 0.06 0.09 0 0.05 0.06 0.07 0.08 0.08 0.09 0 0.07 01 oh. ph p٦ rl m rl rg HEH(0;1) HEH(0;1) 0.14 HEH(0;1) 0.14 0.14 0.12 0.12 0.12 0.1 0.1 0.1 theta 0.08 0.08 theta 0.08 theta 0.06 HEH(0;1) 0.06 0.06 0.04 0.04 0.04 0.02 0.02 0.02 0 0.05 0.06 0.07 0.00 0.09 0.1 0 0.05 0.06 0.07 0.08 0.09 0.1 0 0.05 0.06 0.07 0.08 0.09 0.1 J(th=0, rg=1, rd=1) Jith=0, rg=1, rd=1) ..... J(m=0, rg=1, rd=1)



Parameter values: H = 100,000;  $\delta = 0.2$ ; f = 200; F = 200;  $\overline{p} = 0.05$  for base scenario. f = 100; F = 300 for "High F" and f = 300; F = 100 for "High f".

## C Different size regions

In this appendix we describe the outcome of our model when introducing different size regions. For this we introduce a slight change of notation. We denote by  $M_i$   $(i \in L, H)^{31}$ the stock of houses in each region assuming that  $M_L > M_H$ . The steps to analyze the model are the same as in the text. Hence, we first present the equilibrium candidates, then the conditions that must hold for (Nash) equilibrium. Finally, we present the situation of the reinsurance company and the resulting social cost analysis.

Table C-1 summarizes the equilibrium candidates. The only notable change is the fact that a  $HEL(r_S = 0; r_G = 0)$  is now *ex ante* possible.

Equilibria are obtained by determining the conditions under which a single insurer does not have an incentive to deviate. Table C-2 presents these conditions.

These conditions are very similar to the ones for equally sized regions. Thus one can again determine the area of equilibrium for each candidate, and identify sequences of equilibrium configurations for increases in the premium rate. Similarly, the expressions for the expected profits of the reinsurer are almost identical to the ones in Table 3. Note particularly, that these profits are again non-positive in all cases.

In order to illustrate the social cost of the system under different size regions we specified two sequences a) and b). We assume that  $M_L$  is ten times bigger (1,000,000 houses) than  $M_H$  (100,000 houses), while the other basic parameters are identical to the scenarios before:  $\overline{p} = 0.05$ , f = F = 200,  $\gamma = 0.2$ .

Sequence a) corresponds to sequence ii) in the illustration before, i.e. a small difference in damage probability across regions. We assumed  $p_L = 0.049$  and  $p_H = 0.06$ . Similarly, sequence b) represents sequence iii) from above with  $p_L = 0.04$  and  $p_H = 0.15$ .

Figure 7 in the text presents the social cost for these sequences.

Equilibrium	Number of ins.	Number of ins.	Number of
Equinorium	in $L$ -region	in $H$ -region	generalists
$SE(r_L = 0; r_H = 0)$	$rac{M_L \theta \gamma}{(f+F)}$	$rac{M_H  heta \gamma}{(f\!+\!F)}$	
$SE(r_L = 1; r_H = 0)$	$\frac{M_L(\theta - p_L)}{(f+F)}$	$\frac{M_H \theta \gamma}{(f+F)}$	
$SE(r_L = 1; r_H = 1)$	$rac{M_L(\theta - p_L)}{(f + F)}$	$\frac{M_H(\theta - p_L)}{(f + F)}$	
$GE\left(r_G=0\right)$			$\frac{(M_L+M_H)\theta\gamma}{(2f+F)}$
$GE\left(r_G=1\right)$			$\frac{M_L(\theta - p_L) + M_H(\theta - p_H)}{(2f + F)}$
$HEL(r_S=0; r_G=0)$	$\frac{M_L \theta \gamma}{(f+F)}$		$\frac{M_H \theta \gamma}{f}$
$HEL\left(r_S=1;r_G=0\right)$	$\frac{M_L(\theta - p_L)}{(f + F)}$		$\frac{M_H \theta \gamma}{f \left(2 - \frac{\theta \gamma}{\theta - p_L}\right) + F \left(1 - \frac{\theta \gamma}{\theta - p_L}\right)}$
$HEL(r_S=1;r_G=1)$	$rac{M_L(\theta - p_L)}{(f + F)}$		$\frac{M_H(\theta - p_H)}{f}$
$HEH\left(r_{S}=0;r_{G}=1\right)$		$\frac{M_H\theta\gamma}{(f\!+\!F)}$	$\frac{M_L(\theta - p_L)}{f\left(2 - \frac{\theta - p_H}{\theta\gamma}\right) + F\left(1 - \frac{\theta - p_H}{\theta\gamma}\right)}$

## Table C-1 Equilibrium candidates with different size regions

 $^{31}M$  for "maison".

Table C-2 Summary of Nash equilibrium and minimum number of insurer conditions with different size regions

Equilibrium Candidates	$r_D$	Nash-equilibrium condition
$SE(r_L = 0; r_H = 0)$	0	$\frac{M_L\theta\gamma}{M_L\theta\gamma+f+F}\left(f+F\right) - f \le 0$
$SE(r_L = 1; r_H = 0)$ Case 1	0	$\frac{M_L \theta \gamma}{M_L (\theta - p_L) + f + F} \left( f + F \right) - f \le 0$
$SE(r_L = 1; r_H = 0)$ Case 2	1	$f\left[\frac{M_L(\theta-p_L)}{M_L(\theta-p_L)+f+F} + \frac{(\theta-p_H)}{\theta\gamma} - 2\right] + F\left[\frac{M_L(\theta-p_L)}{M_L(\theta-p_L)+f+F} + \frac{(\theta-p_H)}{\theta\gamma} - 1\right] \le 0$
$SE(r_L = 1; r_H = 1)$	1	$\frac{M_L(\theta - p_L)}{M_L(\theta - p_L) + f + F} \left( f + F \right) - f \le 0$
$\begin{array}{c} GE\left(r_{G}=0\right)\\ \text{Case 1} \end{array}$	0	$f\left[\frac{2M_L}{M_L+M_H}-1\right] + F\left[\frac{M_L}{M_L+M_H}-1\right] < 0$
$\begin{array}{l} GE\left(r_{G}=0\right)\\ \text{Case 2} \end{array}$	1	$f\left[\frac{2M_L}{M_L+M_H}\frac{(\theta-p_L)}{\theta\gamma} - 1\right] + F\left[\frac{M_L}{M_L+M_H}\frac{(\theta-p_L)}{2\theta\gamma} - 1\right] \le 0$
$GE\left(r_{G}=1\right)$	1	$ \begin{aligned} f\left[\frac{2M_L(\theta-p_L)}{M_L(\theta-p_L)+M_H(\theta-p_H)}-1\right] \\ +F\left[\frac{M_L(\theta-p_L)}{M_L(\theta-p_L)+M_H(\theta-p_H)}-1\right] \leq 0 \end{aligned} $
$HEL\left(r_S=0;r_G=0\right)$	0	always Nash
$HEL\left(r_S=1;r_G=0\right)$	0	$f - \frac{\theta\gamma}{\theta - p_L} \left( f + F \right) \le 0$
$HEL\left(r_{S}=1;r_{G}=1\right)$	1	always Nash
$HEH\left(r_{S}=0;r_{G}=1\right)$	1	$f - \frac{(\theta - p_H)}{\theta \gamma} \left( f + F \right) \le 0$

# **CESifo Working Paper Series**

(for full list see www.cesifo.de)

- 1241 Hans Jarle Kind, Karen Helene Midelfart, Guttorm Schjelderup, Corporate Tax Systems, Multinational Enterprises, and Economic Integration, July 2004
- 1242 Vankatesh Bala and Ngo Van Long, International Trade and Cultural Diversity: A Model of Preference Selection, July 2004
- 1243 Wolfgang Eggert and Alfons J. Weichenrieder, On the Economics of Bottle Deposits, July 2004
- 1244 Sören Blomquist and Vidar Christiansen, Taxation and Heterogeneous Preferences, July 2004
- 1245 Rafael Lalive and Alois Stutzer, Approval of Equal Rights and Gender Differences in Well-Being, July 2004
- 1246 Paolo M. Panteghini, Wide vs. Narrow Tax Bases under Optimal Investment Timing, July 2004
- 1247 Marika Karanassou, Hector Sala, and Dennis J. Snower, Unemployment in the European Union: Institutions, Prices, and Growth, July 2004
- 1248 Engin Dalgic and Ngo Van Long, Corrupt Local Government as Resource Farmers: The Helping Hand and the Grabbing Hand, July 2004
- 1249 Francesco Giavazzi and Guido Tabellini, Economic and Political Liberalizations, July 2004
- 1250 Yin-Wong Cheung and Jude Yuen, An Output Perspective on a Northeast Asia Currency Union, August 2004
- 1251 Ralf Elsas, Frank Heinemann, and Marcel Tyrell, Multiple but Asymmetric Bank Financing: The Case of Relationship Lending, August 2004
- 1252 Steinar Holden, Wage Formation under Low Inflation, August 2004
- 1253 Ngo Van Long and Gerhard Sorger, Insecure Property Rights and Growth: The Roles of Appropriation Costs, Wealth Effects, and Heterogeneity, August 2004
- 1254 Klaus Wälde and Pia Weiß, International Competition, Slim Firms and Wage Inequality, August 2004
- 1255 Jeremy S. S. Edwards and Alfons J. Weichenrieder, How Weak is the Weakest-Link Principle? On the Measurement of Firm Owners' Control Rights, August 2004
- 1256 Guido Tabellini, The Role of the State in Economic Development, August 2004

- 1257 François Larmande and Jean-Pierre Ponssard, EVA and the Controllability-congruence Trade-off: An Empirical Investigation, August 2004
- 1258 Vesa Kanniainen and Jenni Pääkkönen, Anonymous Money, Moral Sentiments and Welfare, August 2004
- 1259 Panu Poutvaara and Andreas Wagener, Why is the Public Sector More Labor-Intensive? A Distortionary Tax Argument, August 2004
- 1260 Lars P. Feld and Stefan Voigt, Making Judges Independent Some Proposals Regarding the Judiciary, August 2004
- 1261 Joop Hartog, Hans van Ophem, and Simona Maria Bajdechi, How Risky is Investment in Human Capital?, August 2004
- 1262 Thomas Eichner and Rüdiger Pethig, Efficient Nonanthropocentric Nature Protection, August 2004
- 1263 David-Jan Jansen and Jakob de Haan, Look Who's Talking: ECB Communication during the First Years of EMU, August 2004
- 1264 David F. Bradford, The X Tax in the World Economy, August 2004
- 1265 Hans-Werner Sinn, Migration, Social Standards and Replacement Incomes. How to Protect Low-income Workers in the Industrialized Countries against the Forces of Globalization and Market Integration, August 2004
- 1266 Wolfgang Leininger, Fending off one Means Fending off all: Evolutionary Stability in Submodular Games, August 2004
- 1267 Antoine Bommier and Bertrand Villeneuve, Risk Aversion and the Value of Risk to Life, September 2004
- 1268 Harrie A. A. Verbon and Lex Meijdam, Too Many Migrants, Too Few Services: A Model of Decision-making on Immigration and Integration with Cultural Distance, September 2004
- 1269 Thomas Eichner and Rüdiger Pethig, Economic Land Use, Ecosystem Services and Microfounded Species Dynamics, September 2004
- 1270 Federico Revelli, Performance Rating and Yardstick Competition in Social Service Provision, September 2004
- 1271 Gerhard O. Orosel and Klaus G. Zauner, Vertical Product Differentiation When Quality is Unobservable to Buyers, September 2004
- 1272 Christoph Böhringer, Stefan Boeters, and Michael Feil, Taxation and Unemployment: An Applied General Equilibrium Approach, September 2004

- 1273 Assaf Razin and Efraim Sadka, Welfare Migration: Is the Net Fiscal Burden a Good Measure of its Economics Impact on the Welfare of the Native-Born Population?, September 2004
- 1274 Tomer Blumkin and Volker Grossmann, Ideological Polarization, Sticky Information, and Policy Reforms, September 2004
- 1275 Katherine Baicker and Nora Gordon, The Effect of Mandated State Education Spending on Total Local Resources, September 2004
- 1276 Gabriel J. Felbermayr and Wilhelm Kohler, Exploring the Intensive and Extensive Margins of World Trade, September 2004
- 1277 John Burbidge, Katherine Cuff and John Leach, Capital Tax Competition with Heterogeneous Firms and Agglomeration Effects, September 2004
- 1278 Joern-Steffen Pischke, Labor Market Institutions, Wages and Investment, September 2004
- 1279 Josef Falkinger and Volker Grossmann, Institutions and Development: The Interaction between Trade Regime and Political System, September 2004
- 1280 Paolo Surico, Inflation Targeting and Nonlinear Policy Rules: The Case of Asymmetric Preferences, September 2004
- 1281 Ayal Kimhi, Growth, Inequality and Labor Markets in LDCs: A Survey, September 2004
- 1282 Robert Dur and Amihai Glazer, Optimal Incentive Contracts for a Worker who Envies his Boss, September 2004
- 1283 Klaus Abberger, Nonparametric Regression and the Detection of Turning Points in the Ifo Business Climate, September 2004
- 1284 Werner Güth and Rupert Sausgruber, Tax Morale and Optimal Taxation, September 2004
- 1285 Luis H. R. Alvarez and Erkki Koskela, Does Risk Aversion Accelerate Optimal Forest Rotation under Uncertainty?, September 2004
- 1286 Giorgio Brunello and Maria De Paola, Market Failures and the Under-Provision of Training, September 2004
- 1287 Sanjeev Goyal, Marco van der Leij and José Luis Moraga-González, Economics: An Emerging Small World?, September 2004
- 1288 Sandro Maffei, Nikolai Raabe and Heinrich W. Ursprung, Political Repression and Child Labor: Theory and Empirical Evidence, September 2004

- 1289 Georg Götz and Klaus Gugler, Market Concentration and Product Variety under Spatial Competition: Evidence from Retail Gasoline, September 2004
- 1290 Jonathan Temple and Ludger Wößmann, Dualism and Cross-Country Growth Regressions, September 2004
- 1291 Ravi Kanbur, Jukka Pirttilä and Matti Tuomala, Non-Welfarist Optimal Taxation and Behavioral Public Economics, October 2004
- 1292 Maarten C. W. Janssen, José Luis Moraga-González and Matthijs R. Wildenbeest, Consumer Search and Oligopolistic Pricing: An Empirical Investigation, October 2004
- 1293 Kira Börner and Christa Hainz, The Political Economy of Corruption and the Role of Financial Institutions, October 2004
- 1294 Christoph A. Schaltegger and Lars P. Feld, Do Large Cabinets Favor Large Governments? Evidence from Swiss Sub-Federal Jurisdictions, October 2004
- 1295 Marc-Andreas Mündler, The Existence of Informationally Efficient Markets When Individuals Are Rational, October 2004
- 1296 Hendrik Jürges, Wolfram F. Richter and Kerstin Schneider, Teacher Quality and Incentives: Theoretical and Empirical Effects of Standards on Teacher Quality, October 2004
- 1297 David S. Evans and Michael Salinger, An Empirical Analysis of Bundling and Tying: Over-the-Counter Pain Relief and Cold Medicines, October 2004
- 1298 Gershon Ben-Shakhar, Gary Bornstein, Astrid Hopfensitz and Frans van Winden, Reciprocity and Emotions: Arousal, Self-Reports, and Expectations, October 2004
- 1299 B. Zorina Khan and Kenneth L. Sokoloff, Institutions and Technological Innovation During Early Economic Growth: Evidence from the Great Inventors of the United States, 1790 – 1930, October 2004
- 1300 Piero Gottardi and Roberto Serrano, Market Power and Information Revelation in Dynamic Trading, October 2004
- 1301 Alan V. Deardorff, Who Makes the Rules of Globalization?, October 2004
- 1302 Sheilagh Ogilvie, The Use and Abuse of Trust: Social Capital and its Deployment by Early Modern Guilds, October 2004
- 1303 Mario Jametti and Thomas von Ungern-Sternberg, Disaster Insurance or a Disastrous Insurance – Natural Disaster Insurance in France, October 2004